

Citation: Adjobi CN, Zahouli JZB, Guindo-Coulibaly N, Ouattara AF, Vavassori L, Adja MA (2024) Assessing the ecological patterns of *Aedes aegypti* in areas with high arboviral risks in the large city of Abidjan, Côte d'Ivoire. PLoS Negl Trop Dis 18(11): e0012647. https://doi.org/10.1371/journal. pntd.0012647

Editor: Olaf Horstick, University of Heidelberg, GERMANY

Received: February 2, 2024

Accepted: October 23, 2024

Published: November 18, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012647

Copyright: © 2024 Adjobi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

RESEARCH ARTICLE

Assessing the ecological patterns of *Aedes aegypti* in areas with high arboviral risks in the large city of Abidjan, Côte d'Ivoire

Claver N. Adjobi^{1,2*}, Julien Z. B. Zahouli^{2,3*}, Négnorogo Guindo-Coulibaly¹, Allassane F. Ouattara^{2,4}, Laura Vavassori^{5,6}, Maurice A. Adja¹

 Laboratoire de Biologie et Santé, UFR Biosciences, Université Félix Houphouët-Boigny, Abidjan, Côte d'Ivoire, 2 Centre Suisse de Recherches Scientifiques en Côte d'Ivoire, Abidjan, Côte d'Ivoire, 3 Centre d'Entomologie Médicale et Vétérinaire, Université Alassane Ouattara, Bouaké, Côte d'Ivoire, 4 Unité de Formation et de Recherche Sciences de la Nature, Université Nangui-Abrogoua, Abidjan, Côte d'Ivoire, 5 Swiss Tropical and Public Health Institute, Allschwill, Switzerland, 6 University of Basel, Basel, Switzerland

* claver.adjobi@csrs.ci (CNA); julien.zahouli@csrs.ci (JZBZ)

Abstract

Background

The city of Abidjan, Côte d'Ivoire has increasingly faced multiple outbreaks of *Aedes* mosquito-borne arboviral diseases (e.g., dengue (DEN) and yellow fever (YF)) during the recent years, 2017–2023. Thus, we assessed and compared *Aedes aegypti* larval and adult population dynamics and *Stegomyia* indices in four urbanized areas with differential arboviral incidences in Abidjan, Côte d'Ivoire.

Methods

From August 2019 to July 2020, we sampled *Aedes* mosquito immatures (larvae and pupae), adults and breeding habitats in Anono and Gbagba with high arboviral incidences and Ayakro and Entente with low arboviral incidences in the Abidjan city, using standardized methods. Sampling was conducted in the peridomestic and domestic (indoors and outdoors) premises during short dry season (SDS), short rainy season (SRS), long dry season (LDS) and long rainy season (LRS). The abdomens and ovaries of *Ae. aegypti* females were examined to determine their blood-meal and parity statuses. *Stegomyia* indices (container index: CI, house index: HI and Breteau index: BI), blood-meal status and parity rates were compared by study sites and seasons and with the World Health Organization (WHO)-established epidemic thresholds.

Results

Overall, *Aedes* and arboviral risk indices were high and similar between the four study areas. In total, 86,796 mosquitoes were identified and dominated by *Ae. aegypti* species (97.14%, 84,317/86,796). The most productive larval breeding habitats were tires, discarded containers and water storage containers. CI, HI, and BI in Anono (22.4%, 33.5% and 89.5), Ayakro (23.1%, 43.8% and 91.0), Entente (15.9%, 24.8% and 48.5) and Gbagba (23.3%, 43.0% and 102.0) were high in the respective study sites. *Stegomyia* indices were

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

higher than the WHO-established epidemic thresholds during any seasons for DEN, and LRS and SRS for YF. The numbers of *Ae. aegypti*-positive breeding sites were higher in the domestic premises (68.0%, 900/1,324) than in the peridomestic premises (32.0%, 424/ 1,324). In the domestic premises, *Ae. aegypti*-positive breeding sites (94.6%, 851/4,360) and adult individuals (93.4%, 856/916) were mostly found outdoors of houses. *Aedes aegypti* adult females were mostly unfed (51.3%, 203/396), followed by blood-fed (22.2%, 88/396), gravid (13.9%, 55/396) and half-gravid (12.6%, 50/396), and had parity rate of 49.7% (197/396) that was comparable between the study sites.

Conclusions

The city of Abidjan, Côte d'Ivoire is highly infested with *Ae. aegypti* which showed comparable ecological patterns across study sites and seasons. Thus, the local communities are exposed to high and permanent risks of transmission of DEN and YF viruses that were above the WHO-established epidemic thresholds throughout. The results provide a baseline for future vector studies needed to further characterize the observed patterns of local *Ae. aegypti* abundances and behaviors, and risks of transmission of these arboviruses. Community-based larval source management of identified productive containers might reduce *Ae. aegypti* numbers and risks of transmission of *Aedes*-borne arboviruses in Abidjan, and other sub-Saharan African cities.

Author summary

As most sub-Saharan African cities, Abidjan in Côte d'Ivoire has faced recently a considerable increase in the outbreaks of dengue (DEN) and yellow fever (YF). However, critical data are still lacking on the ecology of the main vector Aedes aegypti and the risk of transmission of DEN and YF viruses. We assessed the ecology of Ae. aegypti mosquitoes and the risk of DEN and YF virus transmission in areas with high and low DEN and YF occurrences in Abidjan. Our findings revealed a significant presence of Ae. aegypti, indicating a high risk of YF and DEN transmission across all study areas. Ae. aegypti larvae were mostly breeding in tires, discarded containers and water storage containers. Meanwhile, elderly adult females were abundant, alongside a significant presence of both unfed hostseeking and blood-fed individuals in and around residences. We observed seasonal risk patterns, with permanent and high threats for DEN observed throughout both rainy and dry seasons over the entire year, alongside elevated risks of YF during both short and long rainy seasons across all study areas. Our findings provide insights into the ecology of Ae. *aegypti* and the epidemiology of DEN and YF, crucial for strategically targeting and controlling this vector in areas both affected and unaffected by the disease outbreaks. Community-based intervention programs for managing identified larval breeding sites might reduce Ae. aegypti numbers to prevent future outbreaks of DEN and YF in Abidjan and other sub-Saharan African cities.

Introduction

Aedes mosquito-borne arboviral diseases such as dengue (DEN), yellow fever (YF), chikungunya (CHIK) and Zika (ZIK), pose a significant threat to over 831 million people, representing 70% of population in sub-Saharan Africa [1]. These diseases have the heaviest public health and socio-economic impacts in urbanized cities [2,3]. Moreover, there is an ongoing resurgence and geographical expansion of arboviral diseases [1,4], intensified by a rapid urbanization, climate change and international mobility and trade [5,6]. West Africa, including Côte d'Ivoire, is one of most important emerging and re-emerging foci and hotspot of arboviruses in Africa [7,8]. In the West African region, over 27,000 arboviral cases were reported between 2007 and 2020, with the highest incidences and greatest burdens observed in major capital cities [7]. In 2023, 171,991 suspected cases of DEN, including 70,223 confirmed and probable cases and 753 deaths have been reported from 15 African countries [9]. A neighboring country of Côte d'Ivoire, Burkina Faso, stands out as the most impacted country with 146,878 suspected cases and 688 deaths [9]. In the absence of licensed vaccines for most arboviruses (except for the YF vaccine) and the lack of widespread routine prophylactic programs for controlling Ae. vectors, surveillance of the primary vector, Ae. aegypti is crucial for preventing, controlling, responding to, and preparing for to prevent arboviral outbreaks. Ae. aegypti, a key vector of arboviruses in Africa [1], can transmit over 5 viruses to humans [10] and exhibits highly anthropophagic behavior, dwelling in and around human habitats such as domestic and peridomestic premises where females predominantly feed on humans and breed in manmade container [9]. The ecological adaptability of Ae. aegypti allows this species to colonize various breeding sites in close proximity to human dwellings [11].

Since 1898 up to the present year (2024), Côte d'Ivoire has experienced multiple outbreaks of YF and DEN, with a notable resurgence and increase incidence in recent years, particularly 2017 to the present [12-18]. Despite the historical and present backgrounds, arboviruses remain uncontrolled, with ongoing resurgence of outbreaks posing a significant public health concern, particularly evident in the densely populated and highly urbanized city of Abidjan, Côte d'Ivoire. The urban environments of Abidjan are permissive to Ae. aegypti. The city is marked by rapid, uncontrolled urbanization and complex land cover changes, driven by poor urban planning and limited environmental and sanitation management services. As a result, Abidjan harbors large numbers of Ae. aegypti (~100% of Aedes mosquitoes) and larval habitats, and has often faced multiple outbreaks of arboviruses (e.g., DEN and YF) [13]. There are currently no specific programs for the routine controls of arboviruses and their vectors. This is largely due to restricted of financial investments, along with limited operational resources and technical capacities. Moreover, Ae. aegypti in Abidjan are resistant to most insecticides used for their control [14]. The government's response to outbreaks, led by the National Institute of Public Hygiene (NIPH) under the Ministry of Health and Public Hygiene (MHPH) of Côte d'Ivoire, primarily relies on sporadic insecticide space spraying targeting adults Aedes mosquitoes, along with and systematic removal, physical destruction and/or treatment of larval breeding sites. The interventions, largely unplanned due to the absence of robust data and accurate predictions, are urgently implemented in response to sudden arboviral outbreaks. However, they frequently yield limited and short-term impacts on local Aedes vector and arboviral control efforts [13–18]. Indeed, the local Ae. aegypti populations recover quickly and arboviruses re-emerge in the intervention areas once the dedicated campaigns are over, as observed in 2017, 2019, 2022, 2023 and 2024 [13-18]. In 2017, Abidjan has recorded outbreaks of DEN (623 suspected, 192 confirmed and 2 fatal cases) [13]. Out of the 192 confirmed DEN cases, 66% were virus serotype 2 (DENV-2), 29% were DENV-3 and 5% were DENV-1. In 2019, outbreaks of DEN (3,201 suspected, 281 confirmed and 2 fatal cases) and YF (89 confirmed and 1 fatal cases) were reported [15,16]. In 2022, Abidjan has faced an outbreak of DEN (181 suspected, 19 confirmed and 1 fatal cases) [16] and in 2023 to an outbreak of DEN that has caused 73 infected cases and 2 deaths [18]. However, reports indicated that arboviral occurrences have shown geographical and seasonal disparities, with the majority (80-90% cases) of cases

being recorded in the health districts of Cocody and Bingerville and during rainy seasons (April-July and September-October), while very few numbers of cases being reported in the other eight health districts of Abidjan city, including Treichville and Yopougon. *Ae. aegypti* populations are highly prevalent (100% of *Aedes* genus) in Abidjan, and larvae mostly breed in discarded items (e.g., cans, tires) and water storage containers [19–21]. The current study aimed at assessing the ecology of the *Ae. aegypti* vector in four different sites with different DEN and YF incidences within the city of Abidjan: Anono in Cocody and Gbagba in Bingerville with high DEN and YF incidence (80–90% cases), and Ayakro in Yopougon and Entente in Treichville with low DEN and YF incidence (<10% cases). We hypothesized that communities are exposed to higher entomological risks of transmission of DEN and YF viruses in Anono and Gbagba (areas with high DEN and YF incidences). We monitored *Ae. aegypti* populations at different development stages (i.e., larvae, pupae and adults) and larval breeding sites in the field using sensitive and standardized methods to test this hypothesis.

Methods

Ethics statement

Before starting the study, the study protocol received ethical approval from the National Ethical Committee (Comité National d'Ethique des Sciences de la Vie et de la Santé) Ministry of Health and Public Hygiene, Côte d'Ivoire (ref: 034-21/MSHP/CNESV5-km). Additionally, authorizations were obtained from the local administrative and health authorities. The local community leaders provided oral informed consent as well. Mosquito collections in households were done with the permission and written informed consent of the owners and/or residents. This study did not involve endangered or protected species.

Study area

The study was conducted in the city of Abidjan (05° 19' N and 4° 01' W) located in southern Côte d'Ivoire (West Africa) (Fig 1). Abidjan is the first and the third largest city of Côte d'Ivoire and South Africa, respectively [22]. The population is estimated at 7 million inhabitants [23]. Abidjan has ten administrative municipalities, including Bingerville, Cocody, Treichville and Yopougon.

Ae. aegypti species and larval breeding sites are highly abundant and ubiquitous in Abidjan [24]. Abidjan has faced multiple outbreaks of DEN and YF. However, the arboviral occurrences and incidences significantly differed from one municipality to another [13–18]. In this study, four urban municipalities were selected based on their epidemiological backgrounds and incidences of DEN and YF: Bingerville (5° 21' N; 3°54' W) and Cocody (5° 20' N; 3° 58' W), located in the Cocody-Bingerville health district where DEN and YF cases are regularly reported; Treichville (05° 19' N; 04° 01' W), situated in the Marcory-Treichville district with few recorded cases of DEN and YF; and Yopougon (5° 20' N; 4° 00' W), where DEN and YF cases are rare. Cocody-Bingerville health district is well known as the main focus arboviral outbreaks of Côte d'Ivoire. Cocody-Bingerville health district accounted for over 80–90% of cases DEN and YF reported between 2017 to 2024, according to hospital data.

Abidjan has a humid and sub-equatorial climate, characterized by four seasons: two rainy seasons from April to July (long rainy season: LRS) and from October to November (short rainy season: SRS) and two dry seasons from December to March (long dry season: LDS) and from August to September (short dry season: SDS). The annual average temperature is around 26–28°C and the annual relative humidity ranges between 75 and 90%. The average annual precipitation ranges between 1000 and 1200 mm.



Map data from openstreetmap.org 2023-20-2023, swiss center for scientific research and our survey



Fig 1. Map showing the location of the study sites in the city of Abidjan, Côte d'Ivoire. The study was conducted in four sites (Anono, Gbagba, Entente and Ayakro) in the city of Abidjan, Côte d'Ivoire. Anono, Gbagba, Entente and Ayakro located in the municipalities of Cocody, Bingerville, Treichville and Yopougon, respectively. As per arboviral epidemic reports from 2017 to 2023, Anono and Gbagba are situated in the health district of Cocody-Bingerville characterized by high arboviral incidence (80–90% dengue and yellow fever cases reported) while Ayakro and Entente are located in the health districts of Yopougon and Treichville characterized by low arboviral incidence (<10% dengue and yellow fever cases reported). The map was created with QGIS software version 3.34 (https://www.qgis.org/), using the basemap is Openstreetmap data so, the basemap is open data, licensed under the Open Data Commons Open Database License (ODbL) by the OpenStreetMap Foundation (OSMF).

https://doi.org/10.1371/journal.pntd.0012647.g001

Study design

In the city of Abidjan, four sites of roughly equal size were selected based on their current arboviral status: Anono (Cocody), Gbagba (Bingerville), Entente (Treichville) and Ayakro (Yopougon). Anono and Gbagba are located in the health district of Cocody-Bingerville, where 80–90% of DEN and YF were recorded, while Ayakro and Entente are located in the health districts of Yopougon and Treichville, respectively, where only a small number of DEN and YF cases were recorded in the recent years (2017–2024). *Aedes* mosquito larvae were sampled among 100 households per study site and per survey. *Aedes* adults were collected from 10 houses in each study site for each survey. For each study site and each collection method, four surveys were carried out, corresponding each to one of the four seasons (i.e., SRS, LDS, LRS and SDS), from August 2019 to July 2020. The collections were done within and around randomly selected households and inside and outside of houses (I.e., main buildings). Surrounding areas within a 25-m radius around the selected households were investigated for *Aedes* larvae and adults. Household was defined as a house occupied by a head and his/her relatives.

If an occupant or owner of a selected household was absent or declined access to the house, this household was replaced by the nearest possible household.

Aedes immature collections

Aedes mosquito immatures (larvae and pupae) were collected in visible and accessible waterholding containers indoors and outdoors among the selected houses. Any water-holding containers were inspected for the presence of *Aedes* immatures (i.e., larvae or pupae). Immatures of Aedes and non-Aedes (e.g., Anopheles, Culex and Lutzia) mosquitoes were collected using flexible rubber tube connected to a manual suction pump, ladles, and pipettes depending of the size of the breeding sites. Larvae of predatory mosquitoes (e.g., Lutzia tigripes) were removed from the samples to avoid predation on Aedes larvae or pupae. Mosquito immatures were filtered through a sieve. They were stored separately per breeding sites within the same water from the inspected larval breeding sites in plastic bags and labelled with information on the household number, study area and collection date. The larval breeding sites were characterized, recording their location (indoors or outdoors) and type (natural or artificial). The breeding sites were classified into five different categories: water storage containers, abandoned containers, tires, natural breeding site and others (e.g., hole of bricks, shoes, tarps, wooden boxes, mortars, metallic sheets). All mosquito immature samples were transferred into a cool box and transported to the insectarium for rearing to adults for morphological identification to species.

Aedes adult collections

Aedes mosquito adults were captured using Prokopack aspirators (Model 1412, John W. Hock Company, USA). Sampling was performed by five well-trained and experienced entomological technicians per survey and the same technicians completed all the surveys. The technicians rotated from one survey to another to minimize biases. The collection box of each Prokopack aspirator was labelled with number of the households and houses sampled, and the initials of the collectors (i.e., technicians), start and end time of sampling, date of sampling and the study area were recorded. Sampling was carried out quarterly in 10 randomly selected houses in the domestic and peridomestic premises. In the domestic premises, sampling was carried out indoors and outdoors of the selected houses. Sampling was done in the morning from 06:00 a. m. to 09:00 a.m. and in the afternoon from 3:00 p.m. to 6:00 p.m. as *Aedes* are diurnal mosquitoes [20]. The time spent in each sampling point (i.e., house and surrounding area) was approximately 30 minutes.

Laboratory procedures

In the insectarium, mosquito immatures were separated and placed up to 20 individuals per 200-ml plastic cup to prevent overcrowding and minimize morality during the rearing. The larvae were fed with fish food in the morning between 07:00 a.m. and 08:00 a.m. Emerged and field-collected pupae were kept until the emergence of adults. All emerged and field-collected adult mosquitoes were identified to species under a binocular magnifying glass using the morphological determination keys based on the color of the scutum, the appearance of the legs and wings, the shape of the thorax and the proboscis of the specimens [25–27]. Field-collected *Ae. aegypti* adult females' abdomen was examined, and females were classified as unfed, blood-fed, half-gravid and gravid individuals according to their abdomen aspect and blood-meal development stage [28]. The ovaries of field-collected *Ae. aegypti* females were dissected and the females were separated as parous or nulliparous individuals. Females were considered parous

when the ovary tracheoles were completely uncoiled, and nulliparous when the ovary tracheoles were coiled into pelota [29].

Data analysis

All statistical analyses were conducted using R Studio version 4.2.0. Significance level of 5% was set for statistical testing. The proportion of positive breeding sites were calculated as the percentage of the number of breeding sites infested with *Ae. aegypti* larvae or pupae relative the total number of inspected water-holding breeding containers. The Z-test was used to compare the proportions of *Ae. aegypti*-positive breeding sites between the study sites and across the seasons.

The potential entomological risk of transmission of DEN and YF viruses was based on *Ae. aegypti* larval indices and assessed through the standard *Stegomyia* indices, including container index (CI), house index (HI) and Breteau index (BI). CI was equal to the percentage of *Ae. aegypti*-positive containers among the total number of water-holding containers inspected. A breeding site was considered positive when it contained at least one *Aedes* larva or pupa. HI was defined as the percentage of houses hosting at least one *Ae. aegypti*-positive container among the total number of households inspected. A house was considered positive when it contained at least one positive breeding site. BI was expressed as the number of *Ae. aegypti*positive containers for 100 houses inspected. CI, HI and BI were compared between the four study sites and between seasons using generalized linear model (GLM to take into account possible interactions between the variables "study site", "season" or "breeding site" with poisson family.

The entomological risk of transmission of YF virus was interpreted according to the World Health Organization (WHO)-established YF epidemics thresholds [30]:

- if CI < 3%: epidemic risk is low; $3\% \le$ CI $\le 20\%$: risk is moderate, and CI > 20%: epidemic risk is high.
- if HI < 4%: risk of an epidemic is low, $4\% \le$ HI \le 35%: risk of an epidemic is moderate. if HI > 35%: risk of an epidemic is high.
- if BI < 5: risk of an epidemic is low. 5 ≤ BI ≤ 50: risk is moderate. BI > 50: risk of an epidemic is high.

The entomological risk of transmission of DEN virus was defined and interpreted according to the Pan American Health Organization (PAHO)-established thresholds [31]:

- if CI > 3% or HI > 4% and BI > 5: risk of an epidemic is high.
- if HI < 0.1%, risk of an epidemic is low.
- if $0.1\% \le HI \le 5\%$, the risk of an epidemic is medium.
- if HI > 5%, the risk of an epidemic is high.

For the field-collected *Ae. aegypti* adults, *Aedes* mean number was expressed as the number of *Aedes* adult specimens per house and per hour (AHH). AHH was tested using counting measure approaches in GLMs to consider possible interactions between the variables "study site", "season" or "house" with poisson family. When over-dispersion was found the negative binomial family structure was used. Repeated measures approach in GLM framework was used because *Aedes* mosquitoes were repeatedly sampled in the same sampling location (house) over time (season). The proportions of *Ae. aegypti* unfed, blood-fed, half-gravid and gravid females were calculated as the respective percentages of unfed, blood-fed, half-gravid

and gravid females (numerator) relative to the total number of females of the same species (denominator). The parity rate was defined as the percentage of parous females (numerator) among total number of females with ovaries dissected (denominator). The proportions of unfed, blood-fed, half-gravid and gravid females and parity rates were analyzed using the Z-test. The means of unfed, blood-fed, half-gravid and gravid and gravid female proportions and parity rates were compared between the study sites using GLM.

Results

Mosquito species composition

Table 1 shows the species composition of mosquitoes collected as immatures (larvae and pupae) and adults in Anono, Ayakro, Entente and Gbagba. A total of 86,796 mosquitoes (46,498 females and 40,298 males) was identified in all four study sites. Mosquitoes belonged to four genera (Aedes 97.15%, n = 84,319), Culex (2.68%, n = 2,322), Anopheles (0.16%, n = 136) and Lutzia (0.02%, n = 19), and seven species dominated by Ae. aegypti (97.14%, 84,317/86,796). Culex quinquefasciatus (67.16%, 1,965/2,926) was the most commonly found species in the adult collections, while Ae. aegypti represented 31.31% (916/2,926) of the fieldcollected adult mosquito fauna. Ae. aegypti (99.44%, 83,401/83,870) strongly dominated the culicid fauna obtained among the larval collections. The highest proportions of mosquito were recorded in Gbagba (30.43%, 26,414/86,796), followed by Ayakro (26.37%, 22,890/86,796), Anono (24.07%, 20,887/86,796) and Entente (19.13%, 16,605/86,796). Aedes genus dominated the mosquito fauna with overall proportion of 97.15% (n = 84,319), and in each study area with 98.11% (20,492/20,887) in Anono, 97.27% (16,152/16,605) in Entente, 97.15% (25,660/ 26,414) in Gbagba, and 96.18% (22,015/22,890) in Ayakro. Ae. aegypti was most abundant mosquito species in the four study sites (97.14%, n = 84,317). Ae. aegypti was the only Aedes species identified in the study sites, except for Gbagba where two additional specimens of Aedes palpalis species (0.01%, 2/25,658) were found. Other medically important non-Aedes species, such as Culex quinquefasciatus (2.65%, 2,298/86,796) a vector of arboviruses and Anopheles gambiae s.l. (0.16%, 136/86,796), a vector of Plasmodium spp, were also collected across the four study areas, though in relatively low proportions.

Aedes aegypti immatures

Breeding sites. Table 2 displays the abundances of larval breeding sites of *Ae. aegypti* found in Anono, Ayakro, Entente and Gbagba across the seasons. In all the four study areas, a total 6,144 potential larval breeding containers were identified, with 21.5% (1,324/6,144) being positive for *Ae. aegypti* larvae. The proportions of *Ae. aegypti*-positive breeding sites varied from one site to another, with the highest proportions found in Gbagba (28.47%, 1,749/6,144), followed by Anono (26.03%, 1,599/6,144), Ayakro (25.65%, 1,576/6,144) and Entente (19.86%, 1,220/6,144). The proportions of *Ae. aegypti*-positive breeding sites were higher in the peridomestic premises (31.9%, 424/1,328) compared with the domestic premises (18.7%, 900/4,816) (S1 Table). The domestic premises (900/1324, 68.0%) harbored higher numbers of *Ae. aegypti*-positive breeding sites than the peridomestic premises (424/1324, 32.0%). In the domestic premises, the majority of *Ae. aegypti*-positive larval breeding sites was found outdoors (19.5%, 851/4,360), while only small proportions were observed indoors (10.7%, 49/456) (S2 Table). Similarly, the numbers of *Ae. aegypti*-positive breeding sites were higher outdoors (851/900, 94.6%) than indoors (49/900, 5.4%).

GLMs indicated that *Ae. aegypti*-positive breeding sites were significantly different between seasons (F = 6.23, df = 3, p = 0.0009). The proportion of *Ae. aegypti* positive breeding sites did not differ statistically between LDS and SDS (Estimate = -0.42 ± 0.34 , z = 1.21, p = 0.22).

Study site	Genus	Species	Immatur	es			Adults				Total	7		
			Female	Male	Total	%	Female	Male	Total	%	Female	Male	Total	%
Anono	Aedes	Ae. aegypti	11119	9175	20294	99.79	81	117	198	35.93	11200	9292	20492	98.11
		Sub-total	11119	9175	20294	99.79	81	117	198	35.93	11200	9292	20492	98.11
	Anopheles	An. gambiae s.l.	6	1	7	0.03	0	0	0	0.00	6	1	7	0.03
		Sub-total	6	1	7	0.03	0	0	0	0.00	6	1	7	0.03
	Culex	Cx. quinquefasciatus	18	17	35	0.17	164	189	353	64.07	182	206	388	1.86
		Sub-total		17	35	0.17	164	189	353	64.07	182	206	388	1.86
	Total		11143	9193	20336	100	245	306	551	100	11388	9499	20887	100
Ayakro	Aedes	Ae. aegypti	11670	10137	21807	99.67	102	106	208	20.59	11772	10243	22015	96.18
		Sub-total	11670	10137	21807	99.67	102	106	208	20.59	11772	10243	22015	96.18
	Anopheles	An. gambiae s.l.	28	24	52	0.24	12	3	15	1.49	40	27	67	0.29
		Sub-total	28	24	52	0.24	12	3	15	1.49	40	27	67	0.29
	Culex	Cx. quinquefasciatus	10	9	19	0.09	408	379	787	77.92	418	388	806	3.52
		Sub-total	10	9	19	0.09	408	379	787	77.92	418	388	806	3.52
	Lutzia	Lu. tigripes	0	2	2	0.01	0	0	0	0.00	0	2	2	0.01
		Sub-total	0	2	2	0.01	0	0	0	0.00	0	2	2	0.01
	Total	Fotal		10172	21880	100	522	488	1010	100	12230	10660	22890	100
Entente	Aedes	Ae. aegypti	8564	7336	15900	99.56	105	147	252	39.69	8669	7483	16152	97.27
		Sub-total	8564	7336	15900	99.56	105	147	252	39.69	8669	7483	16152	97.27
	Anopheles	An. gambiae s.l.	18	15	33	0.21	0	0	0	0.00	18	15	33	0.20
		Sub-total	18	15	33	0.21	0	0	0	0.00	18	15	33	0.20
	Culex	Cx. auinauefasciatus	16	11	27	0.17	157	226	383	60.31	173	237	410	2.47
		Sub-total	16	11	27	0.17	157	226	383	60.31	173	237	410	2.47
	Lutzia	Lu tigrities	7	3	10	0.06	0	0	0	0.00	7	3	10	0.06
	Luiziu	Sub-total	7	3	10	0.06	0	0	0	0.00	7	3	10	0.06
	Total	oub totui	8605	7365	15970	100	262	373	635	100	8867	7738	16605	100
Ghaoha	Aedes	Ae aegypti	13491	11909	25400	98.89	104	154	258	35 34	13595	12063	25658	97.14
Gougou	Titues	Ae palpalis	0	0	0	0.00	1	1	2	0.27	1	1	25050	0.01
		Sub-total	13491	11909	25400	98.89	105	155	260	35.62	13596	12064	25660	97.15
	Anotheles	An gambiae s l	11	14	25 100	0.10	3	1	4	0.55	14	15	29	0.11
	Inophetes	Sub-total	11	14	25	0.10	3	1	4	0.55	14	15	29	0.11
	Culer	Cr. cinereus	0	0	0	0.00	7	4	11	1.51	7	4	11	0.04
	Guida	Cx. nebulosus	0	0	0	0.00	7	6	13	1.78	7	6	13	0.05
		Cx. auinauefasciatus	135	117	252	0.98	250	192	442	60.55	385	309	694	2.63
		Sub-total	135	117	252	0.98	264	202	466	60.55	399	319	718	2.03
	Lutzia	In tigripes	4	3	7	0.03	0	0	0	0.00	4	3	7	0.03
	Luisia	Sub-total	4	3	7	0.03	0	0	0	0.00	4	3	7	0.03
	Total	Jub total	13641	12043	25684	100	372	358	730	100	14013	12401	26414	100
Overall	Aedes	Ae aegypti	44844	38557	83401	99.44	392	524	916	31 31	45236	39081	84317	97.14
overall	Titues	Ae palpalis	0	0	0	0.00	1	1	2	0.07	1	1	2	0.00
		Sub-total	44844	38557	83401	99 44	393	525	918	31.37	45237	39082	84319	97.15
	Anopheles	An gambiae sl	63	50557	117	0.14	15	1	19	0.65	78	59002	136	0.16
	Thopheles	Sub-total	63	54	117	0.14	15	т Л	10	0.65	78	58	136	0.16
	0.1	Cr cinereus	0	0	0	0.00	7	4	11	0.05	7	4	11	0.01
	11/01	Cr. cinereus	0	0	0	0.00	7	4	11	0.38	7	4	11	0.01
	Culex	Cr. nebulocus	0	10		1 11 1111	17	0	1.2	U.44	1	10	113	10.01
	Culex	Cx. nebulosus	0	0	333	0.40	979	986	1965	67.16	1158	1140	2200	2.65
	Culex	Cx. nebulosus Cx. quinquefasciatus	0 179 179	0 154 154	333	0.40	979 993	986 996	1965	67.16	1158	1140	2298	2.65
	Lutzia	Cx. nebulosus Cx. quinquefasciatus Sub-total	0 179 179	0 154 154 8	333 333 19	0.40 0.40 0.02	979 993	986 996	1965 1989	67.16 67.98	1158 1172	1140 1150 8	2298 2322	2.65 2.68
	Lutzia	Cx. nebulosus Cx. quinquefasciatus Sub-total Lu. tigripes Sub-total	0 179 179 11	0 154 154 8 8	333 333 19	0.40 0.40 0.02 0.42	979 993 0	986 996 0	1965 1989 0	67.16 67.98 0.00	1158 1172 11	1140 1150 8 8	2298 2322 19	2.65 2.68 0.02

 Table 1. Species composition of mosquitoes collected as immatures and adults in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

Table 1. (Continued)

Study site	Genus	Species	Immatures				Adults				Total			
			Female	Male	Total	%	Female	Male	Total	%	Female	Male	Total	%
Total			45097	38773	83870	100	1401	1525	2926	100	46498	40298	86796	100

%: percentage. Immatures represent adult mosquitoes emerged from field-collected immatures (larvae and pupae). Adults represent adult field-collected mosquitoes.

https://doi.org/10.1371/journal.pntd.0012647.t001

However, Aedes breeding positivity was significantly higher in LDS compared with LRS (Estimate = 0.85 ± 0.23 , z = 2.6, p = 0.0009) and in LDS compared with SRS (Estimate = 0.85 ± 0.32 , z = 2.62, p = 0.008). The numbers of positive breeding sites were higher in Anono (22.4%, n = 358) than Entente (15.9%, n = 194) (Estimate = -0.80 ± 0.36 , z = -2.23, p = 0.02), but no significant difference was found between Anono and Ayakro (Estimate = -0.05 ± 0.36 , z = -0.14, p = 0.88), and Gbagba (Estimate = -0.06 ± 0.35, z = -0.19, p = 0.84). In Gbagba, the significantly lowest proportions of positive breeding sites of *Ae. aegypti* were observed in SDS (F = 3.98, df = 3, p = 0.03). Conversely, in the three other study sites, no significant differences in Ae. *aegypti*-positive breeding site was noticed between seasons (all p > 0.05). From all potential breeding sites, water storage containers (66.13%, n = 4,063) were the most predominant, followed by tires (20.56%, n = 1,263), discarded containers (8.74%, n = 537), other container categories (4.12%, n = 253) and natural breeding sites (0.46%, n = 28) (χ^2 = 34.17, df = 4, p < 0.0001) (Table 2). Overall, Ae. aegypti colonized all the categories of breeding sites. Aedes aegypti larvae were found in water storage containers (15.01%, 610/4063), tires (31.67%, 400/ 1263), discarded containers (40.59%, 218/537), natural breeding sites (14.29%, 4/28) and other container categories (36.36%, 92/253).

On the 1,324 *Ae. aegypti*-positive breeding sites, water storage containers (46.07%, n = 610) were the most prevalent, followed by tires (30.21%, n = 400), discarded containers (16.46%, n = 218), the other container categories (6.95%, n = 92) and natural breeding sites (0.30%, n = 4) (Z-test χ^2 = 1121.2, df = 4, p < 0.001). The positive breeding sites were found among other containers (4.2%, 15/358), discarded containers (9.8%, 35/358), water storage containers (29.9%, 107/358) and tires (56.1%, 201/358) in Anono. Water storage containers were found to be frequently positive in Ayakro (62.9%, 229/364), Entente (54.6%, 106/194) and Gbagba (41.2%, 168/408) seconded by tires in Ayakro (21.4%, 78/364) and discarded containers in Entente (19.6%, 38/194) and Gbagba (26.0%, 106/408).

In both Anono and Gbagba, the main *Aedes* breeding sites was discarded containers. In Anono, discarded containers were followed by the other container category (35.71%, 15/42), tires (28.47%, 201/706) and water storage containers (13.91%, 107/769), and tires (40.72%, 90/221), other containers (31.82%, 42/132) and water storage containers (14.65%, 168/1147) in Gbagba. In Ayakro, the main *Aedes* positive breeding sites were tires (45.88%, 78/170), followed by other containers (41.86%, 18/43) and discarded containers (38.61%, 39/101). In Entente, of the total *Aedes* breeding sites collected, other containers (47.22%, 17/36) were the most *Aedes*-positive breeding sites, followed by discarded containers (30.89%, 38/123).

In Anono, *Aedes*-positive breeding sites were mostly found in SRS (23.09%, 154/667), followed by LDS (22.84%, 66/289), LRS (22.65%, 106/468) and SDS (18.29%, 32/175) (Table 2). In contrast, in Ayakro, Gbagba and Entente, LRS had the most abundant *Aedes*-positive breeding sites, followed by SRS. The lowest proportion of *Aedes*-positive breeding sites was recorded in SDS in Gbagba (13.77%, 34/247) and Entente (7.34%, 13/177) and LDS in Ayakro (13.98%, 58/415).

Immature productivity. <u>Table 3</u> presents the abundance of *Ae. aegypti* immatures (larvae and pupae) across the four study sites, seasons and larval breeding site categories. Out of

Study	dy Breeding site		SRS			LDS			LRS			SDS				Total					
site	-	N	n	PW	РР	N	n	PW	PP	N	n	PW	РР	N	n	PW	РР	N	n	PW	РР
Anono	Water storage container	315	58	18.4	37.7	128	14	10.9	21.2	226	29	12.8	27.4	100	6	6.0	18.8	769	107	13.9	29.9
	Tire	293	71	24.2	46.1	147	45	30.6	68.2	193	59	30.6	55.7	73	26	35.6	81.3	706	201	28.5	56.1
	Discarded container	42	18	42.9	11.7	6	3	50.0	4.5	33	14	42.4	13.2	1	0	0.0	0.0	82	35	42.7	9.8
	Natural breeding site	0	0	na	0.0	0	0	na	0.0	0	0	na	0.0	0	0	na	0.0	0	0	na	0.0
	Others	17	7	41.2	4.5	8	4	50.0	6.1	16	4	25.0	3.8	1	0	0.0	0.0	42	15	35.7	4.2
	Total	667	154	23.1	100.0	289	66	22.8	100.0	468	106	22.6	100.0	175	32	18.3	100.0	1599	358	22.4	100.0
Ayakro	Water storage container	384	89	23.2	61.0	366	43	11.7	74.1	331	66	19.9	55.9	180	31	17.2	73.8	1261	229	18.2	62.9
	Tire	58	32	55.2	21.9	28	7	25.0	12.1	59	31	52.5	<mark>26</mark> .3	25	8	32.0	19.0	170	78	45.9	21.4
	Discarded container	38	12	31.6	8.2	12	4	33.3	6.9	46	21	45.7	17.8	5	2	40.0	4.8	101	39	38.6	10.7
	Natural breeding site	1	0	0.0	0.0	0	0	na	0.0	0	0	na	0.0	0	0	na	0.0	1	0	0.0	0.0
	Others	27	13	48.1	8.9	9	4	44.4	6.9	5	0	0.0	0.0	2	1	50.0	2.4	43	18	41.9	4.9
	Total	508	146	28.7	100.0	415	58	14.0	100.0	441	118	26.8	100.0	212	42	19.8	100.0	1576	364	23.1	100.0
Entente	Water storage container	330	43	13.0	4.9	184	19	10.3	82.6	236	35	14.8	47.3	136	9	6.6	69.2	886	106	12.0	54.6
	Tire	63	11	17.5	13.1	4	2	50.0	8.7	72	16	22.2	21.6	27	2	7.4	15.4	166	31	18.7	16.0
	Discarded container	75	21	28.0	25.0	14	1	7.1	4.3	28	16	57.1	21.6	6	0	0.0	0.0	123	38	31.0	19.6
	Natural breeding site	4	1	25.0	1.19	3	0	0.0	0.0	1	0	0.0	0.0	1	1	100.0	7.7	9	2	22.2	1.0
	Others	13	8	61.5	9.5	4	1	25.0	4.3	12	7	58.3	9.5	7	1	14.3	7.7	36	17	47.2	8.8
	Total	485	84	17.3	100.0	209	23	11.0	100.0	349	74	21.2	100 <mark>.0</mark>	177	13	7.3	100.0	1220	194	15.9	100.0
Gbagba	Water storage container	321	55	17.1	43.0	302	27	8.9	45.0	373	67	18.0	36.0	151	19	12.6	55.9	1147	168	14.6	41.2
	Tire	73	29	39.7	22.7	59	20	33.9	33.3	73	34	46.6	18.3	16	7	43.8	20.6	221	90	40.7	22.1
	Discarded container	71	33	46.5	25.8	23	11	47.8	18.3	115	56	48.7	30.1	22	6	27.3	17.6	231	106	45.9	26.0
	Natural breeding site	2	0	0.0	0.0	2	0	0.0	0.0	9	2	22.2	1.1	5	0	0.0	0.0	18	2	13.3	0.5
	Others	16	11	68.8	8.6	12	2	16.7	3.3	51	27	52.9	14.5	53	2	3.7	5.9	132	42	31.8	10.3
	Total	483	128	26.5	100.0	398	60	15.1	100.0	621	186	30.0	100.0	247	34	13.7	100.0	1749	408	23.3	100.0
Overall	Water storage container	1350	245	18.1	47.9	980	103	10.5	49.8	1166	197	16.9	40.7	567	65	11.5	53.7	4063	610	15.0	46.1
	Tire	487	143	29.4	27.9	238	74	31.1	35.7	397	140	35.3	28.9	141	43	30.5	35.5	1263	400	31.7	30.2
	Discarded container	226	84	37.2	16.4	55	19	34.5	9.2	222	107	48.2	22.1	34	8	23.5	6.6	537	218	40.6	16.5
	Natural breeding site	7	1	14.3	0.2	5	0	0.0	0.0	10	2	20.0	0.4	6	1	16.7	0.8	28	4	14.3	0.3
	Others	73	39	53.4	7.6	33	11	33.3	5.3	84	38	45.2	7.9	63	4	6.3	3.3	253	92	36.4	6.9
	Total	2143	512	23.9	100.0	1311	207	15.8	100.0	1879	484	25.8	100.0	811	121	14.9	100.0	6144	1324	21.5	100.0

Table 2. Seasonal variations of the abundances of the larval breeding sites of *Aedes aegypti* mosquitoes in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

N: number of wet containers inspected, n: *Aedes*-positive containers, PW: Percentage of *Aedes*-positive breeding sites among wet containers, PP: Proportion of each *Aedes*-positive breeding site type among the all *Aedes*-positive breeding site types. PW and PP are expressed as a percentage (%), na: not applicable, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Others is the category of breeding containers composed of hole of brick, shoes, tarp, flower pot, wooden box, mortar, sheet metal. Natural breeding site is composed of water on land, leaf axils, snail shell, tree hole.

https://doi.org/10.1371/journal.pntd.0012647.t002

83,401 immatures of *Aedes*, 25,400 (30.5%) were found in Gbagba, 21,807 (26.1%) in Ayakro, 20,294 (24.3%) in Anono and 15,900 (19.1%) in Entente. GLM showed that no difference in the abundance of *Ae. aegypti* immatures among the four study sites (F = 0.64, df = 3, p = 0.59). The most productive larval habitats were water storage containers in SRS (43.7%, n = 12,611), tires in LDS (44.7%, n = 5,886) and LRS (33.7%, 12,213) and water storage containers in SDS (46.1%, n = 2,402). The abundance of *Ae. aegypti* immatures significantly varied across the seasons (F = 9.34, df = 3, p < 0.0001). The highest number of *Ae. aegypti* immatures was collected in LRS (43.4%, n = 36,195) and the lowest in SDS (6.25%, n = 5,214).

Overall, the abundance of Ae. aegypti immature differed significantly according to the categories of the breeding sites (F = 6.14, df = 4, p = 0.0003). Of all breeding site categories, water storage containers (37.54%, n = 31,305) and tires (37.8%, n = 31,514) were those that produced more Ae. aegypti immatures, followed by discarded containers (19.4%, n = 16,200), others (4.6%, n = 3,798) and natural breeding sites (0.7%, n = 584). In all the four study sites, the main *Ae. aegypti* immature breeding sites were found in LRS and SRS. Conversely, the lowest number of *Ae. aegypti* immatures was found in both LDS and SDS. In Anono (F = 4.04, df = 3, p = 0.007), Gbagba (F = 7.01, df = 3, p < 0.001), Ayakro (F = 4.96, df = 3, p = 0.002) and Entente (F = 612.21, df = 3, p < 0.0001), the abundance of *Ae. aegypti* immatures showed significant difference across seasons (Table 3). The abundance of Ae. aegypti immatures during the LDS was significantly lower than during LRS (Estimates = 0.48 ± 0.21 , z = 2.34, p = 0.01) and significantly higher than in SDS (Estimate = -0.54 ± 0.26 , z = -2.07, p = 0.03). The abundance of Ae. aegypti immatures during the LDS was significantly higher than in SDS in Anono (Estimate = 0.79 ± 0.27 , z = 2.91, p = 0.003) and in Entente (Estimate = -1.15 ± 0.55 , z = -2.1, p = 0.03), but statistically lower than in LRS in Ayakro (Estimate = 0.86 ± 0.3 , z = 2.88, p = 0.003) and Gbagba (Estimate = 0.79 ± 0.27, z = 2.9, p = 0.003).

Stegomyia indices and DEN and YF virus transmission risk

Table 4 shows the seasonal dynamics of *Stegomyia* indices and risk of transmission of DEN and YF viruses in Anono, Ayakro, Entente and Gbagba. Overall, CI, HI and BI were very high with respective values of 21.5%, 36.3% and 82.8, corresponding to the WHO density scale range of [6–7] and suggesting high risk of transmission of both DEN and YF. Gbagba, Ayakro and Anono were all at high risk, with respective WHO density scale ranges [6–8], [6–7] and [5–7], while entente was at the medium risk with the WHO density scale value of 5. The respective values of the overall CI, HI and BI were estimated at 23.3%, 43.0%, 102.0 in Gbagba, 23.1%, 43.5% and 91.0 in Ayakro, 22.4%, 33.5% and 89.5 in Anono, and 15.9%, 24.8% and 48.5 in Entente. In the all four study areas, CI, HI and BI were high and above the WHO-established epidemic thresholds for DEN. The overall CI, HI and BI were above the WHO-established epidemic thresholds for YF in Anono, Ayakro and Gbagba except for Entente, suggesting levels of risk of transmission of these arboviruses were high in Anono, Ayakro and Gbagba and moderate in Entente. Entente had significantly lower CI (Z-test $\chi^2 = 14.08$, df = 3, p = 0.002), HI (Z-test $\chi^2 = 14.28$, df = 3, p = 0.002) and BI (Z-test $\chi^2 = 19.77$, df = 3, p = 0.0001) compared with the three other study sites.

The overall risk of transmission of DEN and YF viruses varied from medium to high levels over the seasons and showed higher values during the rainy seasons, with the WHO density scale ranges of [7-8] in LRS (CI = 25.8%, HI = 47.8% and BI = 121.0) and [6-8] in SRS (CI = 23.9%, HI = 50.5% and BI = 128.0) (Table 4). CI, HI and BI showed seasonal variations all the study sites, as well. They all were higher than the WHO-established epidemic thresholds in any seasons for DEN and in LRS and SRS for YF (Fig 2). The overall CI values were higher during LRS (25.8%), followed by SRS (23.9%), LDS (15.8%) and SDS (14.9%) (Fig 2A). The

Study site	Breeding site	SRS		LDS		LRS		SDS		Total	
		n	%	n	%	n	%	n	%	n	%
Anono	Water storage container	2670	28.6	224	6.0	870	14.1	144	13.1	3908	19.3
	Tire	5182	55.6	3065	82.7	4228	68.6	954	86.9	13429	66.2
	Discarded container	1270	13.6	153	41.3	807	13.1	0	0.0	2230	11.0
	Others	206	2.2	262	7.1	259	4.2	0	0.0	727	3.6
	Total	9328	100.0	3704	100.0	6164	100.0	1098	100.0	20294	100.0
Ayakro	Water storage container	5134	63.2	1640	53.8	3916	44.6	1262	68.0	11952	54.8
	Tire	2332	28.7	720	23.6	3685	42.0	530	28.6	7267	33.3
	Discarded container	535	6.6	557	18.3	1176	13.4	53	2.9	2321	10.6
	Others	125	1.5	132	4.3	0	0.0	10	0.5	267	1.2
	Total	8126	100.0	3049	100.0	8777	100.0	1855	100.0	21807	100.0
Entente	Water storage container	2349	36.3	1332	57.0	2238	34.6	312	50.7	6231	39.2
	Tire	2385	36.8	671	28.7	2356	36.4	63	10.2	5475	34.4
	Discarded container	1176	18.1	92	3.9	1447	22.4	0	0.0	2715	17.1
	Natural breeding site	194	3.0	0	0.0	0	0.0	226	36.7	420	2.6
	Others	376	5.8	243	10.4	426	6.6	14	2.3	1059	6.7
	Total	6480	100.0	2338	100.0	6467	100.0	615	100.0	15900	100.0
Gbagba	Water storage container	2458	50.2	1707	42.0	4365	29.5	684	41.6	9214	36.3
	Tire	1430	29.2	1430	35.2	1944	13.1	539	32.7	5343	21.0
	Discarded container	840	17.1	837	20.6	6850	46.3	407	24.7	8934	35.2
	Natural breeding site	0	0.0	0	0.0	164	1.1	0	0.0	164	0.6
	Others	173	3.5	92	2.3	1464	9.9	16	1.0	1745	6.9
	Total	4901	100.0	4066	100.0	14787	100.0	1646	100.0	25400	100.0
Overall	Water storage container	12611	43.7	4903	37.3	11389	31.5	2402	4 <mark>6.1</mark>	31305	37.5
	Tire	11329	39.3	5886	44.7	12213	33.7	2086	40.0	31514	37.8
	Discarded container	3821	13.3	1639	12.5	10280	28.4	460	8.8	16200	19.4
	Natural breeding site	194	0.7	0	0.0	164	0.5	226	4.3	584	0.7
	Others	880	3.1	729	5.5	2149	5.9	40	0.8	3798	4.6
	Total	28835	100.0	13157	100.0	36195	100.0	5214	100.0	83401	100.0

Table 3. Seasonal variations of the abundances of *Aedes aegypti* immatures stages in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

%: percentage, n: number of larvae, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Others is the category of breeding containers composed of hole of brick, shoes, tarp, flower pot, wooden box, mortar, sheet metal. Natural breeding site is composed of water on the ground, leaf axils, snail shell, tree hole.

https://doi.org/10.1371/journal.pntd.0012647.t003

highest CI values were recorded during LRS in Gbagba (30.0%), Entente (21.2%) and the whole study area (25.1%) and SRS in Ayakro (28.7%) and Anono (23.1%). The lowest CI values were found during LDS in Ayakro (14.0%) and SDS in Entente (7.3%), Gbagba (13.8%), Anono (18.3%) and the whole study area, (14.8%). The overall HI values were higher during SRS (50.5%), followed by LRS (47.8%), LDS (27.5%) and SDS (19.3%) (Fig 2B). The highest HI values were found during rainy seasons, especially during LRS in Gbagba (61.0%) and SRS in Ayakro (60.0%), Anono (48.0%), Entente (39.0%) and the whole study area (50.5%). All the lowest values of CI were observed during SDS throughout, with values of 11.0% in Entente, 15.0% in Anono, 25.0% in Gbagba and 26.0% in Ayakro and 19.3% in the whole study area. BI was higher during SRS (128.0), followed by LRS (121.0), LDS (51.8) and SDS (30.3) in the whole study site (Fig 2C). The highest BI values were recorded during the rainy seasons, LRS in Gbagba (186.0) and SRS in Anono (154.0), Ayakro (146.0), Entente (84.0) and the whole study area (128.0). The lowest BI were observed during SDS in all the study sites, showing

Study site	Season	CI (%)	HI (%)	BI	WHO density scale	Risk level	Risk level			
						Dengue	Yellow fever			
Anono	SRS	23.1	48.0	154.0	6-8	High	High			
	LDS	22.8	27.0	66.0	5-6	High	Medium			
	LRS	22.7	44.0	106.0	6-8	High	High			
	SDS	18.3	15.0	32.0	4-5	High	Medium			
	Total	22.4	33.5	89.5	5-7	High	Medium			
Ayakro	SRS	28.7	60.0	146.0	7-8	High	High			
	LDS	14.0	34.0	58.0	4-6	High	Medium			
	LRS	26.8	55.0	118.0	7-8	High	High			
	SDS	19.8	26.0	42.0	5	High	Medium			
	Total	23.1	43.5	91.0	6-7	High	High			
Entente	SRS	17.3	39.0	84.0	5-7	High	High			
	LDS	11.0	18.0	23.0	4	Hi <mark>gh</mark>	Medium			
	LRS	21.2	31.0	74.0	5-6	High	Medium			
	SDS	7.3	11.0	13.0	3	High	Medium			
	Total	15.9	24.8	48.5	5	High	Medium			
Gbagba	SRS	26.5	55.0	128.0	8	High	High			
	LDS	15.1	31.0	60.0	5-6	High	Medium			
	LRS	30.0	61.0	186.0	8	High	High			
	SDS	13.8	25.0	34.0	4-5	High	Medium			
	Total	23.3	43.0	102.0	6-8	High	High			
Overall	SRS	23.9	50.5	128.0	6-8	High	High			
	LDS	15.8	27.5	51.8	5-6	High	Medium			
	LRS	25.8	47.8	121.0	7-8	High	High			
	SDS	14.9	19.3	30.3	4	High	Medium			
	Total	21.5	36.3	82.8	6-7	High	High			

Table 4. Risk of transmission of dengue and yellow fever viruses in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

%: percentage, CI: Container index, HI: House index, BI: Breteau index, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season, WHO: World Health Organization, Risk levels estimated according to WHO [30, 31]

https://doi.org/10.1371/journal.pntd.0012647.t004

values of 13.0% in Entente, 32.0% in Anono, 34.0% in Gbagba and 42.0% in Ayakro and 30.3% in the whole study area.

Aedes aegypti adults

Abundance. In total, 916 *Ae. aegypti* adults were identified, with higher proportion in Gbagba (28,17%), followed by Entente (27.51%), Ayakro (22.71%), and Anono (21.62%). *Aedes aegypti* populations were composed of 40.91% females and 59.09% males in Anono (n = 198), 40.31% females and 59.69% males in Gbagba (n = 258), 50.96% females and 49.04% males in Ayakro (n = 208) and 41.67% females and 58.33% males in Entente (n = 252).

Overall AHH (mean ± standard error) was of 4.95 ± 0.86 *Aedes*/house/hour in Anono, 5.2 ± 1.06 *Aedes*/house/hour in Ayakro, 6.30 ± 1.32 *Aedes*/house/hour in Entente, and 6.45 ± 1.07 *Aedes*/house/hour in Gbagba (Fig 3). The respective AHHs of females were of 2.02 ± 0.33 , 2.6 ± 0.38 , 2.65 ± 0.68 and 2.62 ± 0.60 female/house/hour in Anono, Ayakro and Entente and Gbagba (Table 5). Males' AHH was 3.85 ± 0.76 , 3.68 ± 0.82 , 2.92 ± 0.57 and, 2.55 ± 0.46 , male/house/hour in Gbagba, Entente, Anono and Ayakro, respectively (Table 5). AHH did not differ significantly between the four study sites for *Ae. aegypti* overall



Fig 2. Seasonal variations of *Aedes aegypti* **immature indices in the study sites within the city of Abidjan, Côte d'Ivoire.** A: Container index (CI), B: House index (HI), C: Breteau index (BI). The blue dotted line represents the dengue epidemic threshold levels and the red dotted line represents the yellow fever epidemic threshold levels are 3% for container index, 4% for house index and 5 for Breteau index [31]. The yellow fever epidemic threshold levels are 20% for container index, 35% for house index and 50 for Breteau index [30]. SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season.

https://doi.org/10.1371/journal.pntd.0012647.g002

populations (F = 0.49, df = 3, p = 0.68), for females (F = 0.38, df = 3, p = 0.76) and for males (F = 0.92, df = 3 p = 0.43).

AHHs of *Ae. aegypti* varied significantly in the four study sites across the seasons (F = 21.79, df = 3, p < 0.001) (Fig 3). The highest AHHs were observed during SRS in Entente (12.3 ± 3.94 *Aedes*/house/hour), followed by Gbagba (10.2 ± 2.5 *Aedes*/house/hour) and Anono (8.6 ± 1.8 *Aedes*/house/hour), except for Ayakro where the maximum AHH (7.5 ± 1.71 *Aedes*/house/hour) was recorded in LRS. Conversely, the lowest AHHs were recorded during SDS in Anono (1.7 ± 0.60 *Aedes*/house/hour) and Gbagba (2.8 ± 0.68 *Aedes*/house/hour) and LDS in Ayakro (2.4 ± 0.78 *Aedes*/house/hour) and Entente (2.5 ± 0.58 *Aedes*/house/hour). AHHs were not statistically different between the four study sites during the same season, SRS (F = 0.46, df = 3, p = 0.71), LDS (F = 0.83, df = 3, p = 0.48), LRS (F = 0.16, df = 3, p = 0.92) (Fig 3). AHHs did not show any statistical differences between the four study sites in the same seasons, nor in the females and neither the males. In contrast, AHHs were significantly different over seasonal variations between the four study sites for *Ae. aegypti* females (F = 13.77, df = 3, p < 0.001) and males (F = 21.12, df = 3, p < 0.001).

Resting behaviors

Aedes aegypti was highly exophilic, with 93.4% (856/916) of individuals collected outdoors of houses (S3 Table). Only 6.6% of individuals was collected indoors of houses. The overall proportions of exophilic individuals varied slightly among the four study areas, varying between 90.4% (188/208) and 95.5% (189/198). Similarly, the proportions of exophilic individuals varied little across the seasons in all the study sites. *Aedes aegypti* showed lowest proportions of 78.7% (59/75) in Ayakro during LRS and highest proportions of 100% (19/19) in Anono during LDR outdoors of houses.



Fig 3. Seasonal abundance of the adult populations of *Aedes aegypti* in the study sites within the city of Abidjan, Côte d'Ivoire. SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE).

https://doi.org/10.1371/journal.pntd.0012647.g003

Study site	Indicator	SRS	LDS	LRS	SDS	Total
Anono	Number	31	10	30	10	81
	Proportion (%)	38.3	12.3	37.0	12.3	100
	AHH (mean \pm SE) (female + male)	8.6 ± 1.80	1.9 ± 0.52	7.6 ± 2.13	1.7 ± 0.59	4.95 ± 0.86
	AHH (mean ± SE) (female)	3.10 ± 0.71	1.0 ± 0.33	3.0 ± 0.83	1.0 ± 0.33	2.02 ± 0.33
	AHH (mean ± SE) (male)	5.5 ± 1.2	0.9 ± 0.27	4.6 ± 1.44	0.7 ± 0.3	2.92 ± 0.57
Ayakro	Number	39	12	41	14	106
	Proportion (%)	36.8	11.3	38.7	13.2	100
	AHH (mean ± SE) (female + male)	7.3 ± 3.54	2.4 ± 0.77	7.5 ± 1.71	3.6 ± 1.10	5.2 ± 1.06
	AHH (mean ± SE) (female)	3.90 ± 2.50	1.20 ± 0.50	4.10 ± 0.70	1.40 ± 0.43	2.65 ± 0.68
	AHH (mean ± SE) (male)	3.4 ± 1.17	1.2 ± 0.38	3.4 ± 1.13	2.2 ± 0.69	2.55 ± 0.45
Entente	Number	49	13	32	11	105
	Proportion (%)	46.7	12.4	30.5	10.5	100
	AHH (mean ± SE) (female + male)	12.3 ± 3.94	7.5 ± 0.58	7.4 ± 2.52	3.0 ± 0.97	6.3 ± 1.32
	AHH (mean ± SE) (female)	4.9 ± 1.75	1.30 ± 0.34	3.20 ± 1.36	1.10 ± 0.41	2.62 ± 0.60
	AHH (mean ± SE) (male)	7.4 ± 2.58	1.20 ± 0.32	4.20 ± 1.34	1.9 ± 0.64	3.68 ± 0.81
Gbagba	Number	34	15	42	13	104
	Proportion (%)	32.7	14.4	40.4	12.5	100
	AHH (mean ± SE) (female + male)	10.2 ± 2.50	3.4 ± 0.77	9.4 ± 2.75	2.8 ± 0.68	6.45 ± 1.07
	AHH (mean ± SE) (female)	3.40 ± 0.85	1.50 ± 0.40	4.20 ± 0.93	1.30 ± 0.40	2.60 ± 0.39
	AHH (mean ± SE) (male)	6.8 ± 1.93	1.9 ± 0.58	5.2 ± 1.86	1.5 ± 0.40	3.85 ± 0.75
Overall	Number	153	50	145	48	396
	Proportion (%)	38.6	12.6	36.6	12.1	100.0
	AHH (mean ± SE) (female + male)	9.6 ± 1.50	2.55 ± 0.33	7.98 ± 1.12	2.78 ± 0.42	5.72 ± 0.54
	AHH (mean ± SE) (female)	3.82 ± 0.78	1.25 ± 0.19	3.62 ± 0.47	1.20 ± 0.19	2.48 ± 0.26
	AHH (mean ± SE) (male)	5.78 ± 0.90	1.3 ± 0.20	4.35 ± 0.71	1.58 ± 0.27	3.25 ± 0.33

Table 5. Seasonal variations of the abundance of *Aedes aegypti* adult populations in the study sites within city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

% percentage, AHH: Aedes adult per house per hour, SE: standard error, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season.

https://doi.org/10.1371/journal.pntd.0012647.t005

Blood-meal development status

Fig 4 indicates the seasonal variations of the blood-meal status of female *Ae. aegypti* adult populations in Anono, Ayakro, Entente and Gbagba. Overall, *Ae. aegypti* females were mostly unfed $(51.3 \pm 2.5\%, 203/396)$, followed by blood-fed $(22.2 \pm 2.1\%, 88/396)$, gravid $(13.9 \pm 1.7\%, 55/396)$ and half-gravid $(12.6 \pm 1.7\%, 50/396)$. The highest proportions of unfed females were found in Entente $(63.8 \pm 4.7\%, 67/105)$, followed by Ayakro $(48.1 \pm 4.9\%, 51/106)$, Gbagba $(47.1 \pm 4.9\%, 49/104)$ and Anono $(44.4 \pm 5.5\%, 81/106)$ (Fig 4 and S4 Table). The proportions of blood-fed females were higher in Ayakro $(28.3 \pm 4.4\%, 30/106)$, followed by Anono $(25.9 \pm 4.9\%, 21/81)$, Gbagba $(19.2 \pm 3.9\%, 20/104)$ and Entente $(16.2 \pm 3.6\%, 17/105)$. The lowest proportions belonged to half-gravid females in Anono $(11.1 \pm 3.5\%, n = 81)$, Ayakro $(9.43 \pm 2.84, n = 106)$, Entente $(7.62 \pm 2.59\%, n = 105)$ and gravid females in Gbagba $(16.35 \pm 3.63, n = 104)$. The proportions of blood-fed (Z-test $\chi^2 = 5.65$, df = 3, p = 0.12), half-gravid (Z-test $\chi^2 = 5.5$, df = 3, p = 0.13) and gravid (Z-test $\chi^2 = 1.53$, df = 3, p = 0.67) females were statistically similar between the four study sites. However, the proportion of unfed females was significantly higher in Entente than in the three other study sites (Z-test $\chi^2 = 9.26$, df = 3, p = 0.020).

In general, the proportions of unfed females were higher during the rainy seasons $(57.2 \pm 4.1\% \text{ in LRS and } 52.9 \pm 4.0\% \text{ in SRS})$ compared with the dry seasons $(42.0 \pm 7.0\% \text{ in LDS and } 37.5 \pm 7.0\% \text{ in SDS})$ (S2 Fig). In contrast, the proportions of blood-fed females were



Fig 4. Blood-meal statuses in adult females of *Aedes aegypti* in the study sites within the city of Abidjan, Côte d'Ivoire. Error bars show the standard error (SE).

https://doi.org/10.1371/journal.pntd.0012647.g004

higher during the dry seasons ($30.0 \pm 6.5\%$ in LDS and $27.1 \pm 6.4\%$ in SDS) than during the rainy seasons ($22.8 \pm 3.5\%$ in LRS and $17.6 \pm 3.1\%$ in SRS). Generally, the seasonality did not significantly influence the proportions of blood-fed females in all the four study sites (Z-test $\chi^2 = 4.28$, df = 3, p = 0.23). Moreover, the proportions of unfed, half-gravid, and gravid females were each comparable among the seasons in each study site (all p > 0.05) (S3, S4, S5 and S6 Figs).

Parity

Table 6 displays the seasonal variations of the parity rate of *Ae. aegypti* populations in Anono, Ayakro, Entente and Gbagba. *Ae. aegypti* parity rate varied across study sites, with rates recorded at 48,1% in Anono, 51,9% in Ayakro, 48,6% in Entente, and 50,0% in Gbagba. The parity rate did not significantly differ between the four study sites (Z-test $\chi^2 = 0.33$, df = 3, p = 0.95). In addition, there was no significant difference in the proportions of parous and nulliparous females throughout (Z- test $\chi^2 = 0.06$, df = 1, p = 0.802). Overall, the parity rate showed significant difference across seasonal variations (Z-test $\chi^2 = 20.95$, df = 3, p = 0.0001). Indeed, the highest parity rates were found during SRS in Ayakro (64.1 ± 7.7%) and Anono (61.3 ± 8.7%) and during LDS in Entente (69.2 ± 12.8%) and Gbagba (66.7 ± 12.2) (S7 Fig). The parity rate was not significantly different across seasonal variations in Anono (Z-test $\chi^2 = 4.0$, df = 3, p = 0.26) and Entente (Ztest $\chi^2 = 6.07$, df = 3, p = 0.10), but showed significant difference over the seasons in Gbagba (Ztest $\chi^2 = 8.29$, df = 3, p = 0.04) and Ayakro (Z-test $\chi^2 = 9.65$, df = 3, p = 0.02).

Discussion

As most sub-Saharan African cities, the highly urbanized city of Abidjan, Côte d'Ivoire has faced multiple outbreaks of DEN coupled with YF cases from 2017 to 2024 [13–18,32–35].

Study site	Indicator		SRS	LDS	LRS	SDS	Total	
Anono	Abundance	Number	31	10	30	10	81	
		Proportion (%)	38.3	12.3	37.0	12.3	100	
		AHH (mean ± SE)	3.10 ± 0.71	1.0 ± 0.33	3.0 ± 0.83	1.0 ± 0.33	2.02±0.33	
	Parity	Parous	19	5	11	4	39	
		Nulliparous	12	5	19	6	42	
		Parity rate (mean ± SE) (%)	61.3 ± 8.7	50.0 ± 15.8	36.7 ± 8.8	40.0 ± 15.5	48.1±5.5	
Ayakro	Abundance	Number	39	12	41	14	106	
		Proportion (%)	36.8	11.3	38.7	13.2	100	
		AHH (mean ± SE)	3.90 ± 2.50	1.20 ± 0.50	4.10 ± 0.70	1.40 ± 0.43	2.65±0.68	
	Parity	Parous	25	6	18	6	55	
		Nulliparous	14	6	23	8	51	
		Parity rate (mean ± SE) (%)	64.1 ± 7.7	50.0 ± 14.4	43.9 ± 7.7	42.9 ± 13.2	51.9±4.9	
Entente	Abundance	Number	49	13	32	11	105	
		Proportion (%)	46.7	12.4	30.5	10.5	100	
		AHH (mean ± SE)	4.90 ± 1.75	1.30 ± 0.34	3.20 ± 1.36	1.10 ± 0.41	2.62±0.60	
	Parity	Parous	19	9	16	7	51	
		Nulliparous	30	4	16	4		
		Parity rate (mean ± SE) (%)	38.8 ± 7.0	69.2 ± 12.8	50.0 ± 8.8	63.6 ± 14.5	48.6±4.9	
Gbagba	Abundance	Number	34	15	42	13	104	
		Proportion (%)	32.7	14.4	40.4	12.5	100	
		AHH (mean ± SE)	3.40 ± 0.85	1.50 ± 0.40	4.20 ± 0.93	1.30 ± 0.40	2.60 ± 0.39	
	Parity	Parous	21	10	14	7	52	
		Nulliparous	13	5	28	6	48	
		Parity rate (mean ± SE) (%)	61.8 ± 8.3	66.7 ± 12.2	33.3 ± 7.3	53. <mark>8 ±</mark> 13.8	50.0 ± 4.9	
Overall	Abundance	Number	153	50	145	48	396	
		Proportion (%)	38.6	12.6	36.6	12.1	100.0	
		AHH (mean ± SE)	3.82 ± 0.78	1.25 ± 0.19	3.62 ± 0.47	1.20 ± 0.19	2.48 ± 0.26	
	Parity	Parous	84	30	59	24	197	
		Nulliparous	69	20	86	24	199	
		Parity rate (mean ± SE) (%)	54.9 ± 4.0	6 <mark>0.0 ±</mark> 6.9	40.7 ± 4.1	50.0 ± 7.2	49.7 ± 2.5	

Table 6. Seasonal variations of the abundance and the parity status of female populations of *Aedes aegypti* in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020.

%: percentage, AHH: Aedes adult per house per house, SE: standard error, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season.

https://doi.org/10.1371/journal.pntd.0012647.t006

Therefore, it is crucial to better understand the extent to which *Ae. aegypti* and arbovirusrelated epidemic risks spread across Abidjan. Thus, this study assessed and compared the ecology of *Ae. aegypti* and the risk of transmission of DEN and YF viruses in Abidjan, among Anono, Ayakro, Gbagba and Entente characterized by geographical differences in reported arboviral incidences. Anono and Gbagba are located in the health district of Cocody-Bingerville that accounted for 80–90% reported cases of DEN and YF, while only few cases of arboviral diseases (<10%) were recorded in Ayakro and Entente. The results showed that all the four study sites were heavily infested with *Ae. aegypti*, resulting in medium to high levels of risk of DEN and YF virus transmission. Overall, no significant differences were observed in the ecological patterns of local *Ae. aegypti* populations, and *Stegomyia* indices (CI: 21.2%, HI: 36.3% and BI: 82.8) found to be higher than the WHO-established epidemic thresholds. These findings suggest that the populations are exposed to very high and similar threats from DEN and YF outbreaks in the study sites, potentially across the large city of Abidjan, Côte d'Ivoire.

Our data showed that Ae. aegypti, the main arbovirus vector [8], was the most abundant mosquito species (>97%) and almost the only Aedes species in all the study sites. The dominance of Ae. aegypti found in our study sites is consistent with that recorded traditionally in large African urbanized cities where Ae. albopictus are absent or rare, as reported in Abidjan [36] and Ouagadougou, Burkina Faso [36]. Its high abundance among larval collections may be explained by the fact that this study was conducted in the urban area of Abidjan. The dominance of this vector in urban areas of Africa is well documented in several studies [37,38]. Similarly, its predominance among the mosquitoes collected during the larval survey can also be explained by the types of containers inspected (domestic or abandoned containers, tires, etc.). These are preferential breeding sites for this vector [39,24]. Aedes albopictus, another key arbovirus vector species previously notified in Abidjan in 2010 [40] and 2014 [19], was not sampled in our present study. Although the presence of Ae. albopictus has been reported in Abidjan, its absence in our mosquito samples could be attributed to its establishment in an environment where Ae. aegypti was already well established, which may have led to its gradual disappearance. According to Hashim et al. [41], to avoid competition, Ae. aegypti tends not to lay its eggs in sites already colonized by Ae. aegypti and vice versa. This indicates that these two species have difficulties cohabiting in the same breeding site [41]. Moreover, studies have shown that Ae. aegypti is better adapted to urban environments, whereas Ae. albopictus prefers peridomestic habitats with denser vegetation [42,43]. Ae. aegypti is a highly anthropophilic species and its high abundance observed in the current study may be explained by the massive presence of humans offering large opportunities of blood-feeding for females and high numbers of unmanaged discarded and water storage containers acting as suitable breeding sites for ovipositing, all provided by rapid, uncontrolled and unplanned urbanization [19,21,44].

Our data displayed that Ae. aegypti-positive larval breeding sites in the four study areas were highly abundant and diversified, with strong proportions of discarded containers, tires and water storage containers. In Africa, the key container habitats with highest numbers of Ae. *aegypti* pupae and/or larvae are discarded car tires, large domestic water containers (drums and barrels) and small containers (including discarded vessels) in Burkina Faso [45], and jerricans, drums, used or discarded containers and tires in Kenya [46]. In all our study areas, the breeding site positivity was permanently high across the seasons. The most productive larval habitats were water storage containers in SRS, tires in LDS and LDS and water storage containers in SDS. The proportions of *Aedes*-positive breeding sites were directly linked to water storage practices in the domestic areas and rainfalls in the peri-domestic premises, as previously reported in Côte d'Ivoire [19] and Puerto Rico [47]. We observed in the domestic premises that the local populations collected and/or stored water for long duration for various house tasks (e.g., cleaning, cooking, washing, bathing, building and watering plants and animals), to prevent water shortages. People stored potable water for long period, even in the dry season to deal with water interruptions or limited access [47]. This might allow Ae. aegypti females to lay their eggs into the water containers that result in the emergence and proliferation of adults [19]. The *Aedes* eggs laid during the dry season could resist to desiccation, and remain viable, and hatch during the next rainy season, thus resulting in an increase in the numbers of larvae and adults [48]. The high Aedes-positivity of tires indicates a high proportion of tires infested with Ae. aegypti immatures. Indeed, the vehicle tires were used for producing of the local dish "Attiéké", roofing, and decoration in domestic areas, and sold or abandoned in the markets and at roadsides in the peridomestic areas [19]. The high abundances of tires and discarded containers might be attributed to the poor management of solid or plastic waste and the lack of community awareness [39,49-51]. Unmanaged tires and discarded cans are more stable as they are less subjected to human disturbance and pressure due to a poor environmental sanitation service. Tires are suitable breeding sites for Ae. aegypti larvae due to their ability to hold

water and preserve water for long time and provide shade [39,49,50]. Water in tires is rich in organic detritus and microbial organisms that are an adequate food source for rapid development [50]. Additionally, the temperature, humidity and hiddenness inside tires create a favorable environment for the best development of *Aedes* larvae to pupae, and then the proliferation of adult populations [39]. Overall, the high presence of breeding sites correlated with high abundances of *Ae. aegypti* immature and adults of in the four study sites [52].

Our results demonstrated that all the study sites were exposed to high and similar risks of transmission of DEN and YF viruses, although the epidemics are generally localized and restricted to the health district of Cocody-Bingerville where Anono and Gbagba are located. Moreover, the potential entomological risk indices were above the WHO-established epidemic thresholds in the four study areas. The potential risk of the emergence of a DEN epidemic remained high whatever the seasons and the study sites, while the risk of YF outbreaks was high in the rainy seasons, and medium or low in dry seasons in all the study areas. The high values of the Aedes larval and arboviral risk indices suggest that the entire Abidjan city is possibly exposed to large epidemic threats, even if sporadic DEN and YF epidemics have appeared only in some places so far. Indeed, although the risk of an epidemic was medium or low, it was more than sufficient for an epidemic of DEN or YF to occur in all the study areas as an outbreak may occur even if the epidemic risk index is lower than the WHO-established threshold [30]. Indeed, local populations continued to create and maintain *Aedes* breeding sites, despite awareness campaigns as a part of the public health responses. These awareness campaigns were a part of public health responses and were conducted under the aegis of the MHPH through the NIPH [9]. The campaigns consisted of sensitizing, mobilizing and engaging the local populations with the supports of political, religious and community leaders for managing, destroying, removing, or insecticide-treating solid and plastic waste serving as Aedes breeding grounds (discarded tires, cans, etc.) and cleaning and covering piped water storage recipients [24,48]. For the 2017-outbreak responses, over 17,000 households were inspected and over 250,000 Aedes larval breeding sites were eliminated and/or treated with insecticides [9]. Stegomyia indices were very high and statistically comparable between the study areas, independently from the difference in the numbers of reported DEN and YF cases. Therefore, actual epidemic and epidemic-free zones should be included into the arbovirus vector surveillance and control programs.

Our results showed that AHHs of Ae. aegypti did not statistically differ between the four study sites. To our experience, AHHs were potentially high (5.72 Aedes/house/hour), demonstrating the strong anthropophilic habits of local Ae. aegypti. Resting and blood-fed females were mostly collected outdoors in all the study sites, probably due to their diurnal activities, and exophagic and exophilic habits [53–57]. The increased number of unfed females suggests a surge in swarming or reproductive activity of Ae. aegypti populations during the rainy season. This highlights a clear correlation between Ae. aegypti abundance and rainfall patterns, as reported by several authors [46,58]. Newly emerged females are thus more likely to seek blood meals for their initial egg-laying cycles, which heightens the risk of arbovirus transmission. This pattern aligns with the frequent occurrence of epidemics, typically observed during the rainy season [59]. The high numbers of Ae. aegypti adults and parous females could be attributable to the environmental and biological characteristics of urban areas that might be favorable to their survival and longevity [53,54]. Their high presence in and around houses within the domestic premises may be due to their strong anthropophilic behaviors, and this could increase the risk of DEN and YF virus transmission to people [53,54]. The current and previous studies did not analyze Ae. aegypti blood-meals and hosts in Abidjan city and Côte d'Ivoire. However, a study reported that the Ae. aegypti human-blood index (HBI) was higher than 90% in a similar city, Ouagadougou, Burkina Faso a neighboring country [45]. The close

proximity and short distances between of the larval breeding sites and human residencies may increase human-*Aedes* vector contacts, and human-biting, blood-feeding, resting and egg-laying opportunities [60]. Thus, the almost anthropophily of *Ae. aegypti* found here could increase of risks of DEN and YF virus transmission to humans, mainly outdoors of houses in all the study areas.

Our study suggested that the Stegomyia indices were not predictive of current patterns of DEN and YF outbreaks, probably due to some limitations that should be addressed. Indeed, no significant differences were found in Ae. aegypti larval indices and adult numbers among study areas, despite differences in the numbers of DEN and YF cases. This suggests that the high abundance of Ae. aegypti and high epidemic risk indices alone could not be enough to produce an outbreak of DEN or YF. Similarly, previous studies reported no correlations between entomological risk indices and arboviral epidemics in urbanized cities of Burkina Faso [45] and Kenya [46]. Additional investigations using new methods or new tools are needed to address these limitations for better understanding the differential occurrences of DEN and YF cases in Abidjan. This includes, among others, serological diagnostic tests through rapid diagnostic tests (RDTs) or real-time reverse transcriptase PCR (rRT-PCR) for the detection of DEN and YF within the local population. Indeed, we did not analyze the Ae. *aegypti* bloodmeal sources to determine hosts or reservoirs and vector competence for and infection with DEN and YF viruses due to logistical and funding limitations. Moreover, as the differences in arboviral incidences were captured only among clinical cases from hospitals [61], assessing arboviral infections with arboviruses in the whole populations in the study areas are needed to determine the true prevalences. Indeed, arboviral burden is underestimated as infections are often misdiagnosed as malaria, recorded as non-malarial acute febrile illnesses or unidentified fevers due to a lack of technical capacities [61,62]. Some socio-epidemiological factors such as local community culture, beliefs, knowledge, behaviors, needs and priorities and urban poverty may challenge the diagnostics and vector control efforts. Human movements may compromise the identification of the location of Aedes human-biting and arbovirus transmission places, since Ae, aegypti is a diurnal vector and people can receive arbovirus-infested bites at their work places or schools outside of their residences. Assessment of the urbanization level, habitation type, land-cover type, housing conditions, vegetation, water supply and/or waste management and their interactions with the ecologies of Ae. aegypti and arboviruses is required. Ae. aegypti preference for ovipositing in domestic versus peridomestic, indoor versus outdoor, and water storage versus discarded containers is suggestive of behavioral and/or genetic variations in the vector populations [63], thus calling for further investigations.

The uncontrolled galloping urbanization of Abidjan city has resulted in numerous artificial breeding sites conducive to *Ae. aegypti* development and persistence. Our data are important for *Aedes* vector control moving away from reactive entomological control operations to more proactive preventative control. Indeed, there are still no well-structured programs dedicated for routine diagnostics, surveillance and treatment for most arboviruses due to a critical lack of financial, technical and logistical resources. While *Aedes* vector control is crucial to prevent DEN and YF virus transmission, stand-alone government outbreak responses mainly based on outdoor sporadic space-spraying showed short-term and limited effectiveness for controlling *Aedes* vectors and arboviral outbreaks. We identified that key *Ae. aegypti* larval habitats were water storage containers, and unmanaged waste materials such as tires and discarded containers abundantly dispersed in the public and private places. Multisectoral collaborations involving decision-makers, policy-makers, municipal authorities, local health authorities, urban planners, citizen scientists, community health workers and local community leaders and members, and community-based clean-up campaigns focusing on appropriate information, education and empowerment programs are essential for sustainable management and recycling of

identified larval breeding containers [64,65]. Such a holistic, integrated and inclusive vector management practices may be effective for the sustainable controls of *Aedes* vectors and arboviral epidemics in the study areas, and more widely in the city of Abidjan.

Conclusion

The current study conducted in urban areas with high and low DEN and YF incidences within the city of Abidjan, Côte d'Ivoire showed that all study areas were abundantly infested with *Ae. aegypti* immatures and adults. The key larval breeding sites were water storage receptacles, tires and discarded containers mostly found outside of houses and during the rainy seasons. The *Stegomyia* indices were higher than the WHO-established epidemic thresholds and similar between all the study sites, suggesting that *Ae. aegypti* larval indices were not predictive of DEN and YF outbreaks. These results suggest that local communities were exposed to high biting and resting rates of *Ae. aegypti* and high arbovirus transmission risks outdoors. The outcomes improved our understanding of the distributional patterns of *Aedes* vectors and DEN and YF virus transmission risks in space and time within different eco-epidemiological areas. Overall, the findings offer a baseline for future studies to better understand the relationship between *Ae. aegypti* vectors, the observed risk patterns and DEN and YF incidences for costeffective prevention of these diseases. In the meantime, a community-based larval source management of identified productive containers might reduce *Ae. aegypti* numbers and risks of transmission of arboviruses in Abidjan, and more widely in other sub-Saharan African cities.

Supporting information

S1 Fig. The abundance of *Aedes aegypti* adult populations in the study sites within the city of Abidjan, Côte d'Ivoire. Error bars show the standard error (SE). (TIF)

S2 Fig. Seasonal variations in blood-meal statuses in adult females of *Aedes aegypti* in all the study sites. SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S3 Fig. Seasonal variations in blood-meal statuses in adult females of *Aedes aegypti* **in the study site of Anono, Côte d'Ivoire.** SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S4 Fig. Seasonal variations in blood-meal statuses in adult females of *Aedes aegypti* in the **study site of Ayakro, Côte d'Ivoire.** SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S5 Fig. Seasonal variations in blood-meal statuses in adult females of *Aedes aegypti* in the **study site of Entente, Côte d'Ivoire.** SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S6 Fig. Seasonal variations in blood-meal statuses in adult females of *Aedes aegypti* **in the study site of Gbagba, Côte d'Ivoire.** SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S7 Fig. Seasonal variation in the parity rates of *Aedes aegypti* **in the study sites within Abidjan, Côte d'Ivoire.** SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Error bars show the standard error (SE). (TIF)

S1 Table. Seasonal variations of the abundance of the larval breeding sites of *Aedes aegypti* in domestic and peridomestic premises in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020. WSC: Water storage containers, DC: Discarded containers, NBS: Natural breeding sites, N: number of wet containers inspected, n: *Aedes*-positive containers, PW: Percentage of *Aedes*-positive breeding sites among wet containers, PP: Proportion of each *Aedes*-positive breeding site type among the all Aedes-positive breeding site types. PW and PP are expressed as a percentage (%), na: not applicable, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Others is the category of breeding containers composed of hole of brick, shoes, tarp, flower pot, wooden box, mortar, sheet metal. Natural breeding site is composed of water on land, leaf axils, snail shell, tree hole.

(DOCX)

S2 Table. Seasonal variations of the abundance of the larval breeding sites of *Aedes aegypti* collected indoors and outdoors of houses in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020. WSC: Water storage containers, DC: Discarded containers, NBS: Natural breeding sites, N: number of wet containers inspected, n: *Aedes*-positive containers, PW: Percentage of *Aedes*-positive breeding sites among wet containers, PP: Proportion of each *Aedes*-positive breeding site type among the all *Aedes*-positive breeding site types. PW and PP are expressed as a percentage (%), na: not applicable, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. Others is the category of breeding containers composed of hole of brick, shoes, tarp, flower pot, wooden box, mortar, sheet metal. Natural breeding site is composed of water on land, leaf axils, snail shell, tree hole. (DOCX)

S3 Table. Seasonal variations of Aedes aegypti adults collected outdoors and indoors of houses in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020. %: Percentage, n: Number of collected adult Aedes aegypti mosquitoes.
(DOCX)

S4 Table. Seasonal variations of the blood-meal status of female populations of *Aedes aegypti* in the study sites within the city of Abidjan, Côte d'Ivoire from August 2019 to July 2020. %: percentage, n: number of *Aedes aegypti* mosquitoes, SE: standard error, SRS: short rainy season, LDS: long dry season, LRS: long rainy season, SDS: short dry season. (DOCX)

S1 Data. Data of Assessing the ecological patterns of Aedes aegypti in areas with high arboviral risks in the large city of Abidjan, Côte d'Ivoire. (XLSX)

Acknowledgments

The authors would like to thank the administrative authorities for authorizing research in the various communities, the community leaders, the customary authorities and the residents of Anono, Gbagba, Ayakro and Entente.

Author Contributions

Conceptualization: Claver N. Adjobi, Julien Z. B. Zahouli, Maurice A. Adja.

Data curation: Claver N. Adjobi, Julien Z. B. Zahouli, Allassane F. Ouattara.

Formal analysis: Claver N. Adjobi, Allassane F. Ouattara.

Funding acquisition: Julien Z. B. Zahouli.

Investigation: Claver N. Adjobi, Julien Z. B. Zahouli.

Methodology: Claver N. Adjobi, Julien Z. B. Zahouli, Négnorogo Guindo-Coulibaly, Maurice A. Adja.

Project administration: Claver N. Adjobi, Julien Z. B. Zahouli.

Resources: Julien Z. B. Zahouli.

Software: Claver N. Adjobi, Allassane F. Ouattara.

Supervision: Julien Z. B. Zahouli, Maurice A. Adja.

Validation: Claver N. Adjobi, Julien Z. B. Zahouli, Maurice A. Adja.

Visualization: Claver N. Adjobi, Julien Z. B. Zahouli, Maurice A. Adja.

Writing - original draft: Claver N. Adjobi.

Writing – review & editing: Claver N. Adjobi, Julien Z. B. Zahouli, Négnorogo Guindo-Coulibaly, Laura Vavassori, Maurice A. Adja.

References

- Weetman D, Kamgang B, Badolo A, Moyes CL, Shearer FM, Coulibaly M, et al. Aedes Mosquitoes and Aedes-Borne Arboviruses in Africa: Current and Future Threats. Int J Environ Res Public Health. 2018 Jan 28; 15(2):220. https://doi.org/10.3390/ijerph15020220 PMID: 29382107; PMCID: PMC5858289.
- Kraemer MUG, Faria NR, Reiner RC Jr, Golding N, Nikolay B, Stasse S, et al. Spread of yellow fever virus outbreak in Angola and the Democratic Republic of the Congo 2015–16: a modelling study. Lancet Infect Dis. 2017 Mar; 17(3):330–338. https://doi.org/10.1016/S1473-3099(16)30513-8 Epub 2016 Dec 23. Erratum in: Lancet Infect Dis. 2019 Apr;19(4):e109. PMID: 28017559; PMCID: PMC5332542.
- Lee JS, Mogasale V, Lim JK, Ly S, Lee KS, Sorn S, et al. A multi-country study of the economic burden of dengue fever based on patient-specific field surveys in Burkina Faso, Kenya, and Cambodia. PLoS Negl Trop Dis. 2019 Feb 28; 13(2):e0007164. https://doi.org/10.1371/journal.pntd.0007164 PMID: 30817776; PMCID: PMC6394908.
- Braack L, Gouveia de Almeida AP, Cornel AJ, Swanepoel R, de Jager C. Mosquito-borne arboviruses of African origin: review of key viruses and vectors. Parasit Vectors. 2018 Jan 9; 11(1):29. <u>https://doi.org/10.1186/s13071-017-2559-9</u> PMID: 29316963; PMCID: PMC5759361.
- Kraemer MUG, Reiner RC Jr, Brady OJ, Messina JP, Gilbert M, Pigott DM, et al. Past and future spread of the arbovirus vectors *Aedes aegypti* and *Aedes albopictus*. Nat Microbiol. 2019 May; 4(5):854–863. https://doi.org/10.1038/s41564-019-0376-y Epub 2019 Mar 4. Erratum in: Nat Microbiol. 2019 Mar 21;: Erratum in: Nat Microbiol. 2019 Apr 8;: PMID: 30833735; PMCID: PMC6522366.
- Mordecai EA, Ryan SJ, Caldwell JM, Shah MM, LaBeaud AD. Climate change could shift disease burden from malaria to arboviruses in Africa. Lancet Planet Health. 2020 Sep; 4(9):e416–e423. https://doi. org/10.1016/S2542-5196(20)30178-9 PMID: 32918887; PMCID: PMC7490804.
- Buchwald AG, Hayden MH, Dadzie SK, Paull SH, Carlton EJ. Aedes-borne disease outbreaks in West Africa: A call for enhanced surveillance. Acta Trop. 2020 Sep; 209:105468. <u>https://doi.org/10.1016/j.actatropica.2020.105468</u> Epub 2020 May 19. PMID: 32416077.
- Agboli E, Zahouli JBZ, Badolo A, Jöst H. Mosquito-Associated Viruses and Their Related Mosquitoes in West Africa. Viruses. 2021 May 12; 13(5):891. <u>https://doi.org/10.3390/v13050891</u> PMID: <u>34065928</u>; PMCID: PMC8151702.
- 9. World Health Organization (WHO). Dengue- Global situation. 2023 [cited 24 Apr 2024]. Available: https://www.who.int/emergencies/disease-outbreak-news/item/2023-DON498

- Weaver SC, Reisen WK. Present and future arboviral threats. Antiviral Res. 2010 Feb; 85(2):328–45. https://doi.org/10.1016/j.antiviral.2009.10.008 Epub 2009 Oct 24. PMID: 19857523; PMCID: PMC2815176.
- 11. Sene NM, Diouf B, Gaye A, Ndiaye EH, Ngom EM, Gueye A, et al. Blood Feeding Patterns of Aedes aegypti Populations in Senegal. Am J Trop Med Hyg. 2022 Mar 28; 106(5):1402–5. https://doi.org/10. 4269/ajtmh.21-0508 Epub ahead of print. PMID: 35344930; PMCID: PMC9128688.
- Wondji C. La fièvre jaune à Grand-Bassam (1899–1903). In: Revue française d'histoire d'outre-mer, tome 59, n°215, 2e trimestre 1972. pp. 205–239; https://doi.org/10.3406/outre.1972.1596
- 13. WHO. Dengue fever–Côte d'Ivoire. 2017 [cited 9 Aug 2023]. Available: https://www.who.int/ emergencies/disease-outbreak-news/item/04-august-2017-dengue-cote-d-ivoire-en
- Konan LY, Oumbouke WA, Silué UG, Coulibaly IZ, Ziogba JT, N'Guessan RK, et al. Insecticide Resistance Patterns and Mechanisms in *Aedes aegypti* (Diptera: Culicidae) Populations Across Abidjan, Côte d'Ivoire Reveal Emergent Pyrethroid Resistance. J Med Entomol. 2021 Jul 16; 58(4):1808–1816. https://doi.org/10.1093/jme/tjab045 PMID: 33876233.
- 15. World Health Organization (WHO). Weekly Bulletin on Outbreaks and Other Emergencies—Week 42 2019, 14–20 October 2019. Geneva, Switzerland; 2019.
- Carlson R. Urban Yellow Fever Outbreak Infects 89 Causing One Fatality. 2019 [cited 9 Aug 2023]. Available: https://www.vaxbeforetravel.com/republic-côte-divoire-ivory-coast-reports-yellow-fevervirus-outbreak-abidjan
- AfricaNews. Dengue fever outbreak: One dead, 11 cases recorded in Ivory Coast. In: Africanews [Internet]. 4 May 2022 [cited 25 Sep 2022]. Available: https://www.africanews.com/2022/05/04/dengue-fever-outbreak-one-dead-11-cases-recorded-in-ivory-coast/
- Côte d'Ivoire—Dengue outbreak 2023. In: FluTrackers News and Information [Internet]. 14 Jul 2023 [cited 30 Nov 2023]. Available: https://flutrackers.com/forum/forum/emerging-diseases-other-healththreats-alphabetical-a-thru-h/dengue/976705-côte-d-ivoire-dengue-outbreak-2023
- Zahouli JBZ, Koudou BG, Müller P, Malone D, Tano Y, Utzinger J. Urbanization is a main driver for the larval ecology of *Aedes* mosquitoes in arbovirus-endemic settings in south-eastern Côte d'Ivoire. PLoS Negl Trop Dis. 2017 Jul 13; 11(7):e0005751. https://doi.org/10.1371/journal.pntd.0005751 PMID: 28704434; PMCID: PMC5526600.
- Zahouli JBZ, Koudou BG, Müller P, Malone D, Tano Y, Utzinger J. Effect of land-use changes on the abundance, distribution, and host-seeking behavior of *Aedes* arbovirus vectors in oil palm-dominated landscapes, southeastern Côte d'Ivoire. PLoS One. 2017 Dec 7; 12(12):e0189082. https://doi.org/10. 1371/journal.pone.0189082 PMID: 29216248; PMCID: PMC5720743.
- Guindo-Coulibaly N, Kpan MDS, Adja AM, Kouadio AMN, Assouho KF, Zoh DD, et al. Seasonal variation and intra urban heterogeneity of the entomological risk of transmission of dengue and yellow fever in Abidjan, Côte d'Ivoire. Med Vet Entomol. 2022 Sep; 36(3):329–337. https://doi.org/10.1111/mve. 12571 Epub 2022 Mar 30. PMID: 35352845.
- 22. Yao YM, Brenoum DK, Atta LK. Abidjan, Metropole Nationale Et Internationale. Int.jour.sci.res.mana. 2022; 10: 41–52. https://doi.org/10.18535/ijsrm/v10i07.g02
- **23.** Recensement Général de la Population et de l'Habitat (RGPH). Recensement général de la population et de l'habitat 2021. Résultats globaux. 2021.
- 24. Fofana D, Beugré JMV, Yao-Acapovi GL, Lendzele SS. Risk of Dengue Transmission in Cocody (Abidjan, Ivory Coast). J Parasitol Res. 2019 Jan 14; 2019:4914137. https://doi.org/10.1155/2019/4914137 PMID: 30755798; PMCID: PMC6348904.
- 25. Mattingly PF, Bruce-Chwatt LJ. Morphology and bionomics of *Aëdes (Stegomyia) pseudoafricanus* Chwatt (Diptera, Culicidae), with some notes on the distribution of the subgenus stegomyia in Africa. Ann Trop Med Parasitol. 1954 Jun; 48(2):183–93. https://doi.org/10.1080/00034983.1954.11685614 PMID: 13189334.
- Gillies MT, De Meillon B. The Anophelinae of Africa south of the Sahara (Ethiopian zoogeographical region). 2d ed. Johannesburg: South African Institute for Medical Research; 1968. 26. Highton RB. Taxonomic keys for the identification of the afrotropical mosquitoes. 1983.
- 27. Huang Y-M. The subgenus Stegomyia of *Aedes* in the Afrotropical Region with keys to the species (Diptera: Culicidae). Zootaxa. 2004; 700: 1. https://doi.org/10.11646/zootaxa.700.1.1
- Azam FB, Carney RM, Kariev S, Nallan K, Subramanian M, Sampath G, et al. Classifying stages in the gonotrophic cycle of mosquitoes from images using computer vision techniques. Sci Rep. 2023 Dec 13; 13(1):22130. https://doi.org/10.1038/s41598-023-47266-7 PMID: 38092769; PMCID: PMC10719391.
- 29. Detinova TS, Beklemishev WN, Berthram DS. Méthodes à appliquer pour classer par groupe d'âges les diptères présentant une importance médicale notamment certains vecteurs du paludisme. Organistion Mondiale de la Santé, série de monographie N°47; 1963.

- World Health Organization (1971). Technical guide for a system of yellow fever surveillance = guide technique pour l'établissement d'un système surveillance de la fièvre jaune. Weekly Epidemiological Record = Relevé épidémiologique hebdomadaire, 46 (49), 493–500. https://iris.who.int/handle/10665/ 218521
- **31.** Pan American Health Organisation. Dengue and dengue hemorrhagic fever in the Americas: guidelines for prevention and control. Washington DC; 1994.
- Ano AKMN, Coulibaly D, Bénie VJ, Akani BC, Douba A, Ahoussou EM, et al. 411 –Situation épidémiologique de la dengue en Côte d'Ivoire de 2017 à 2020. Revue d'Épidémiologie et de Santé Publique. 2022; 70: S146. https://doi.org/10.1016/j.respe.2022.06.048
- Sylla Y, Diane MK, Adjogoua VE, Kadjo H, Dosso M. Dengue Outbreaks in Abidjan: Seroprevalence and Cocirculating of Three Serotypes in 2017. OSIR. 2021; 14. https://doi.org/10.59096/osir.v14i3. 262527
- World Health Organization (WHO). Weekly bulletin on outbreaks and other emergencies, week 33 : 14– 20 August 2023. Africa Region: Health Emergency Information and Risk Assessment; 2023 p. 18. Available: https://iris.who.int/bitstream/handle/10665/372631/ OEW33-1420082023.pdf
- 35. World Health Organization (WHO). WHO African Region Health Emergency Situation Report. Multicountry Outbreak of Dengue. 2024 Jan p. 11.
- Kpan MDS, Adja AM, Guindo-Coulibaly N, Assouho KF, Kouadio AMN, Azongnibo KRM, et al. Spatial Heterogeneity and Seasonal Distribution of *Aedes (Stegomyia) aegypti* (L) in Abidjan, Côte d'Ivoire. Vector Borne Zoonotic Dis. 2021 Oct; 21(10):769–776. https://doi.org/10.1089/vbz.2021.0002 Epub 2021 Jul 12. PMID: 34252330.
- Guindo-Coulibaly N, Adja AM, Coulibaly JT, Kpan MDS, Adou KA, Zoh DD. Expansion of Aedes africanus (Diptera: Culicidae), a sylvatic vector of arboviruses, into an urban environment of Abidjan, Côte d'Ivoire. J Vector Ecol. 2019 Dec; 44(2):248–255. https://doi.org/10.1111/jvec.12356 PMID: 31729805.
- Ouattara LPE, Sangaré I, Namountougou M, Hien A, Ouari A, Soma DD, et al. Surveys of Arboviruses Vectors in Four Cities Stretching Along a Railway Transect of Burkina Faso: Risk Transmission and Insecticide Susceptibility Status of Potential Vectors. Front Vet Sci. 2019 May 28; 6:140. <u>https://doi.org/ 10.3389/fvets.2019.00140 PMID: 31192232; PMCID: PMC6546915.</u>
- Ferede G, Tiruneh M, Abate E, Kassa WJ, Wondimeneh Y, Damtie D, et al. Distribution and larval breeding habitats of *Aedes* mosquito species in residential areas of northwest Ethiopia. Epidemiol Health. 2018 Apr 23; 40:e2018015. https://doi.org/10.4178/epih.e2018015
 PMID: 29748457; PMCID: PMC5968207.
- 40. Konan YL, Coulibaly ZI, Koné AB, Ekra KD, Doannio JM, Dosso M, et al. Species composition and population dynamics of *Aedes* mosquitoes, potential vectors of arboviruses, at the container terminal of the autonomous port of Abidjan, Côte d'Ivoire. Parasite. 2013; 20:13. https://doi.org/10.1051/parasite/2013013 Epub 2013 Apr 9. PMID: 23567057; PMCID: PMC3718536.
- Hashim NA, Ahmad AH, Talib A, Athaillah F, Krishnan KT. Co-breeding Association of *Aedes albopictus* (Skuse) and *Aedes aegypti* (Linnaeus) (Diptera: Culicidae) in Relation to Location and Container Size. Trop Life Sci Res. 2018 Mar; 29(1):213–227. https://doi.org/10.21315/tlsr2018.29.1.14 Epub 2018 Mar 2. PMID: 29644025; PMCID: PMC5893233.
- 42. Tsuda Y, Suwonkerd W, Chawprom S, Prajakwong S, Takagi M. Different spatial distribution of Aedes aegypti and Aedes albopictus along an urban-rural gradient and the relating environmental factors examined in three villages in northern Thailand. J Am Mosq Control Assoc. 2006 Jun; 22(2):222–8. https://doi.org/10.2987/8756-971X(2006)22[222, DSDOAA]2.0.CO;2. PMID: 17019767.
- Lambrechts L, Scott TW, Gubler DJ. Consequences of the expanding global distribution of Aedes albopictus for dengue virus transmission. PLoS Negl Trop Dis. 2010 May 25; 4(5):e646. https://doi.org/10. 1371/journal.pntd.0000646 PMID: 20520794; PMCID: PMC2876112.
- Agha SB, Tchouassi DP. Urbanization of Aedes mosquito populations and evolution of arboviral disease risk in Africa. Curr Opin Insect Sci. 2022 Dec; 54:100988. <u>https://doi.org/10.1016/j.cois.2022</u>. 100988 Epub 2022 Nov 1. PMID: 36332839.
- 45. Badolo A, Sombié A, Yaméogo F, Wangrawa DW, Sanon A, Pignatelli PM, et al. First comprehensive analysis of *Aedes aegypti* bionomics during an arbovirus outbreak in west Africa: Dengue in Ouagadougou, Burkina Faso, 2016–2017. PLoS Negl Trop Dis. 2022 Jul 6; 16(7):e0010059. <u>https://doi.org/10. 1371/journal.pntd.0010059 PMID: 35793379; PMCID: PMC9321428.</u>
- 46. Agha SB, Tchouassi DP, Bastos ADS, Sang R. Assessment of risk of dengue and yellow fever virus transmission in three major Kenyan cities based on Stegomyia indices. PLoS Negl Trop Dis. 2017 Aug 17; 11(8):e0005858. https://doi.org/10.1371/journal.pntd.0005858 PMID: 28817563; PMCID: PMC5574621.
- Barrera R, Amador M, MacKay AJ. Population dynamics of *Aedes aegypti* and dengue as influenced by weather and human behavior in San Juan, Puerto Rico. PLoS Negl Trop Dis. 2011 Dec; 5(12):e1378.

https://doi.org/10.1371/journal.pntd.0001378 Epub 2011 Dec 20. PMID: 22206021; PMCID: PMC3243685.

- Guindo-Coulibaly N, Adja AM, Koudou BG, Konan YL, Diallo M, Koné AB, et al. Distribution and Seasonal Variation of *Aedes aegypti* in the Health District of Abidjan (Côte d'Ivoire) Eur J Sci Res.2010; 40: 522–30.
- 49. Getachew D, Tekie H, Gebre-Michael T, Balkew M, Mesfin A. Breeding Sites of Aedes aegypti: Potential Dengue Vectors in Dire Dawa, East Ethiopia. Interdiscip Perspect Infect Dis. 2015; 2015;706276. https://doi.org/10.1155/2015/706276 Epub 2015 Sep 7. PMID: 26435712; PMCID: PMC4576013.
- González MA, Rodríguez-Sosa MA, Vásquez-Bautista YE, Rosario EDC, Durán-Tiburcio JC, Alarcón-Elbal PM. A survey of tire-breeding mosquitoes (Diptera: Culicidae) in the Dominican Republic: Considerations about a pressing issue. Biomedica. 2020 Sep 1; 40(3):507–515. https://doi.org/10.7705/ biomedica.5200 PMID: 33030829; PMCID: PMC7666857.
- 51. Hiscox A, Kaye A, Vongphayloth K, Banks I, Piffer M, Khammanithong P, et al. Risk factors for the presence of Aedes aegypti and Aedes albopictus in domestic water-holding containers in areas impacted by the Nam Theun 2 hydroelectric project, Laos. Am J Trop Med Hyg. 2013 Jun; 88(6):1070–8. https://doi.org/10.4269/ajtmh.12-0623 Epub 2013 Mar 4. PMID: 23458958; PMCID: PMC3752805.
- Agha SB, Tchouassi DP, Bastos ADS, Sang R. Dengue and yellow fever virus vectors: seasonal abundance, diversity and resting preferences in three Kenyan cities. Parasit Vectors. 2017 Dec 29; 10 (1):628. https://doi.org/10.1186/s13071-017-2598-2 PMID: 29284522; PMCID: PMC5747025.
- 53. Kamau WW, Sang R, Ogola EO, Rotich G, Getugi C, Agha SB, et al. Survival rate, blood feeding habits and sibling species composition of Aedes simpsoni complex: Implications for arbovirus transmission risk in East Africa. PLoS Negl Trop Dis. 2022 Jan 24; 16(1):e0010171. https://doi.org/10.1371/journal. pntd.0010171 PMID: 35073317; PMCID: PMC8812930.
- 54. Kamau WW, Sang R, Rotich G, Agha SB, Menza N, Torto B, et al. Patterns of *Aedes aegypti* abundance, survival, human-blood feeding and relationship with dengue risk, Kenya. Front Trop Dis. 2023; 4: 1113531. https://doi.org/10.3389/fitd.2023.1113531
- 55. Egid BR, Coulibaly M, Dadzie SK, Kamgang B, McCall PJ, Sedda L, et al. Review of the ecology and behaviour of *Aedes aegypti* and *Aedes albopictus* in Western Africa and implications for vector control. Curr Res Parasitol Vector Borne Dis. 2022; 2:100074. https://doi.org/10.1016/j.crpvbd.2021.100074 PMID: 35726222; PMCID: PMC7612875.
- Manzambi EZ, Mbuka GB, Ilombe G, Takasongo RM, Tezzo FW, Del Carmen Marquetti M, et al. Behavior of Adult Aedes aegypti and Aedes albopictus in Kinshasa, DRC, and the Implications for Control. Trop Med Infect Dis. 2023 Mar 30; 8(4):207. https://doi.org/10.3390/tropicalmed8040207 PMID: 37104333; PMCID: PMC10143671.
- Facchinelli L, Badolo A, McCall PJ. Biology and Behaviour of *Aedes aegypti* in the Human Environment: Opportunities for Vector Control of Arbovirus Transmission. Viruses. 2023 Feb 27; 15(3):636. <u>https://doi.org/10.3390/v15030636 PMID: 36992346</u>; PMCID: PMC10053764.
- 58. Che-Mendoza A, Martin-Park A, Chávez-Trava JM, Contreras-Perera Y, Delfín-González H, González-Olvera G, et al. Abundance and Seasonality of *Aedes aegypti* (Diptera: Culicidae) in Two Suburban Localities of South Mexico, With Implications for *Wolbachia* (Rickettsiales: Rickettsiaceae)-Carrying Male Releases for Population Suppression, *Journal of Medical Entomology*, Volume 58, Issue 4, July 2021, Pages 1817–1825, https://doi.org/10.1093/jme/tjab052
- **59.** Johansson MA, Dominici F, Glass GE. Local and global effects of climate on dengue transmission in Puerto Rico. PLoS Negl Trop Dis. 2009; 3(2):e382. https://doi.org/10.1371/journal.pntd.0000382 Epub 2009 Feb 17. PMID: 19221592; PMCID: PMC2637540.
- Santos Dos Andrade P, Urbinatti PR, da Rocha Coelho R, de Sá Almeida RMM, Ribeiro SS, de Lima-Camara TN. Parity and gonotrophic discordance of females of *Aedes aegypti* and *Aedes albopictus* (Diptera: Culicidae) in the city of São Paulo, SP, Brazil. J Vector Ecol. 2019 Dec; 44(2):233–240. https:// doi.org/10.1111/jvec.12354 PMID: 31729798.
- L'Azou M, Succo T, Kamagaté M, Ouattara A, Gilbernair E, Adjogoua E, et al. Dengue: etiology of acute febrile illness in Abidjan, Côte d'Ivoire, in 2011–2012. Trans R Soc Trop Med Hyg. 2015 Nov; 109 (11):717–22. https://doi.org/10.1093/trstmh/trv076 Epub 2015 Sep 18. PMID: 26385938; PMCID: PMC4603269.
- Stoler J, Al Dashti R, Anto F, Fobil JN, Awandare GA. Deconstructing "malaria": West Africa as the next front for dengue fever surveillance and control. Acta Trop. 2014 Jun; 134:58–65. https://doi.org/10. 1016/j.actatropica.2014.02.017 Epub 2014 Mar 5. PMID: 24613157.
- Rose NH, Badolo A, Sylla M, Akorli J, Otoo S, Gloria-Soria A, et al. Dating the origin and spread of specialization on human hosts in *Aedes aegypti* mosquitoes. Elife. 2023 Mar 10; 12:e83524. <u>https://doi.org/ 10.7554/eLife.83524</u> PMID: 36897062; PMCID: PMC10038657.

- 64. Bonnet E, Fournet F, Benmarhnia T, Ouedraogo S, Dabiré R, Ridde V. Impact of a community-based intervention on *Aedes aegypti* and its spatial distribution in Ouagadougou, Burkina Faso. Infect Dis Poverty. 2020 Jun 5; 9(1):61. https://doi.org/10.1186/s40249-020-00675-6 PMID: 32503665; PMCID: PMC7275586.
- Freeman EA, Carlton EJ, Paull S, Dadzie S, Buchwald A. Utilizing citizen science to model the distribution of *Aedes aegypti* in West Africa. J Vector Ecol. 2022 Jun; 47(1):117–127. https://doi.org/10.52707/ 1081-1710-47.1.117 PMID: 36629363.

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0012647 November 18, 2024